# Spectrum Decision in Cognitive Radio Networks: A Survey

Moshe Timothy Masonta, Student Member, IEEE, Mjumo Mzyece, Member, IEEE, and Ntsibane Ntlatlapa, Senior Member, IEEE

Abstract-Spectrum decision is the ability of a cognitive radio (CR) to select the best available spectrum band to satisfy secondary users' (SUs') quality of service (QoS) requirements, without causing harmful interference to licensed or primary users (PUs). Each CR performs spectrum sensing to identify the available spectrum bands and the spectrum decision process selects from these available bands for opportunistic use. Spectrum decision constitutes an important topic which has not been adequately explored in CR research. Spectrum decision involves spectrum characterization, spectrum selection and CR reconfiguration functions. After the available spectrum has been identified, the first step is to characterize it based not only on the current radio environment conditions, but also on the PU activities. The second step involves spectrum selection, whereby the most appropriate spectrum band is selected to satisfy SUs' OoS requirements. Finally, the CR should be able to reconfigure its transmission parameters to allow communication on the selected band. Key to spectrum characterization is PU activity modelling, which is commonly based on historical data to provide the means for predicting future traffic patterns in a given spectrum band. This paper provides an up-to-date survey of spectrum decision in CR networks (CRNs) and addresses issues of spectrum characterization (including PU activity modelling), spectrum selection and CR reconfiguration. For each of these issues, we highlight key open research challenges. We also review practical implementations of spectrum decision in several CR platforms.

*Index Terms*—Cognitive Radio, Primary User, Reconfiguration, Secondary User, Spectrum Characterization, Spectrum Decision, Spectrum Selection.

#### I. INTRODUCTION

**R** ECENT advancements in wireless technologies, such as software defined radios (SDRs), promise to address some of the major limitations experienced in legacy wireless communication systems. One of these limitations is inefficient utilization and management of the radio frequency (RF) spectrum in both licensed and unlicensed bands. Traditionally, RF spectrum is managed by the regulatory agencies through the assignment of fixed portions of spectrum to individual users in the form of renewable licenses. Although this regulatory

The financial support of the CSIR is gratefully acknowledged.

approach ensures interference-free communications between radio terminals, it suffers from inefficient spectrum utilization. The available literature shows that spectrum utilization, on a block of licensed RF band, varies from 15% to 85% at different geographic locations at a given time [1]–[3]. As the demand for advanced broadband wireless technologies and services increases, traditional static spectrum regulation policies are becoming obsolete. To keep up with growing demand, there is a need for more efficient dynamic spectrum access (DSA) [4] technologies and regulatory approaches.

The need for DSA or opportunistic spectrum access (OSA) was first proposed for the United States by the Federal Communications Commission (FCC) in 2003 [1]. This need was mainly driven by the threat of lack of operating spectrum for future wireless technologies. This move was recently followed by another important decision by the FCC in 2008 [5] and the Office of Communications (Ofcom) in the United Kingdom in 2010 [6], to open up television white spaces (TVWS) for unlicensed utilization. TVWS refers to large portions of RF spectrum, in the very high frequency (VHF) and ultra high frequency (UHF) bands, that will become vacant after the switch-over from analogue to digital TV [7]. Alongside these developments, there has been a strong trend towards research and development of cognitive radio (CR) [8] technology to optimally access the usable spectrum opportunistically and dynamically. A CR is "an intelligent wireless communication system capable of changing its transceiver parameters based on interaction with the external environment in which it operates" [1]. CR is therefore seen as an enabling technology for efficient DSA.

While several approaches are proposed for achieving DSA (such as the dynamic exclusive use model, the spectrum commons model, and the hierarchical access model [4]), our focus in this paper shall be on DSA using CR technology.

The process of realizing efficient spectrum utilization using CR technology requires a dynamic spectrum management framework (DSMF). In this paper, we shall adopt the DSMF proposed in [2] due to its clarity and relevance to our discussion. This DSMF consists of *spectrum sensing, spectrum decision, spectrum sharing* and *spectrum mobility*, as shown in Fig. 1. Spectrum sharing refers to coordinated access to the selected channel by the secondary users (SUs) or CR users. (While the terms "SU" and "CR user" are used interchangeably, in this paper we shall only use the term SU). Spectrum mobility is the ability of a CR to vacate the channel when a licensed user is detected. Spectrum sensing involves identification of spectrum holes and the ability to quickly detect the onset of

Manuscript received 14 October 2011; 1st revision resubmitted March 2012; 2nd revision resubmitted 17 September 2012; manuscript accepted 24 October 2012.

M. T. Masonta is with the Council for Scientific and Industrial Research (CSIR) and also a doctorate candidate at Tshwane University of Technology (TUT), Pretoria, Republic of South Africa (RSA), (e-mail: mmasonta@csir.co.za), M. Mzyece is with the French South African Institute of Technology (F'SATI), Dept. of Electrical Engineering, TUT, Pretoria, RSA, (e-mail: mzyecem@tut.ac.za), and N. Ntlatlapa is also with the CSIR, Pretoria, RSA, (e-mail:nntlatlapa@csir.co.za).



Fig. 1: Dynamic Spectrum Management Framework [2]

licensed or primary user (PU) transmissions in the spectrum hole occupied by the SUs. Spectrum decision refers to the ability of the SUs to select the best available spectrum band to satisfy users' quality of service (QoS) requirements. In this paper we will focus on the spectrum decision component of the DSMF.

Spectrum decision involves three main functions [2]: *spectrum characterization, spectrum selection* and *CR reconfiguration*. Once vacant spectrum bands are identified (using spectrum sensing, geo-location databases or other techniques), each spectrum band is characterized based on local observations and on statistical information of the primary networks (which is normally called PU activities). The second step involves the selection of the most appropriate spectrum band, based on the spectrum band characterization. Thirdly, a CR should be able to reconfigure its transceiver parameters to support communication within the selected spectrum band. The required functions for the spectrum decision framework are summarised in Fig. 2. In order to perform these functions, the following questions need to be answered:

- 1. How can the available spectrum be characterized?
- 2. How can the best spectrum band be selected to satisfy the SU's QoS requirements?
- 3. What is the optimal technique to reconfigure the CR for the selected spectrum band? (And how?)

The above questions form the basis of spectrum characterization, spectrum selection and CR reconfiguration, respectively, as shown in Fig. 2. In this paper, we provide a comprehensive, up-to-date survey of the key research work on spectrum decision in cognitive radio networks (CRNs). We also identify and discuss some of the key open research challenges related to each aspect of the spectrum decision framework. This paper surveys the literature over the period 2003 to mid-2012 on spectrum decision in CRNs. This survey does not cover work done on spectrum sensing, spectrum sharing, spectrum mobility or geo-location databases.

We choose to focus on spectrum decision because of its importance in and centrality to the DSMF in CRNs and because it has received relatively little attention compared to other components of the CR DSMF (namely spectrum sensing, spectrum mobility and spectrum sharing). In many ways, spectrum decision represents the culmination of the DSMF



Fig. 2: Spectrum Decision Framework

in CRNs. We can limit our focus to this one aspect of DSMF based on the well-known communications engineering principles of modularity and abstraction, perhaps most famously and powerfully exemplified in Shannon's 1948 classic paper [9].

The remainder of this paper is arranged as follows. Section II provides the background of CR technology and the motivation for performing spectrum decision in CRNs. Section III outlines CR standardization and regulation activities which are related to spectrum decision in CRNs. Section IV discusses spectrum characterization, the first of the three major spectrum decision functions in CRNs. Section V focuses on spectrum selection, the second major spectrum decision function in CRNs. Section VI covers CR reconfiguration and reconfigurable parameters, the final major spectrum decision function in CRNs. Section VII presents related work on practical implementations of spectrum decision on CR platforms. Future developments in CRNs are reviewed in Section VIII. Section IX concludes the paper.

#### II. OVERVIEW OF CRNs

In this section we provide an overview of CR technology and different CRN topologies. We briefly mention generic problems affecting spectrum decision functions due to the time-varying nature and fluctuations of the available spectrum in CRNs.

# A. Cognitive Radio Overview

Recently, CR has received considerable attention from the research community as an enabling technology for efficient management of RF spectrum. In order to achieve DSA, a CR should be both spectrum and policy agile [10]. A spectrum agile CR is capable of operating over a wide range of frequency spectrum; while a policy agile CR will be aware of the constraints under which it operates (such as the rules for opportunistically using the vacant spectrum bands). Practically, CR builds on the software defined radio (SDR) architecture with added intelligence to learn from its operating environment and adapt to statistical variations in the input stimuli for efficient resource utilization [11]. With the current threat of



Fig. 3: Centralized CRN Topology

spectrum scarcity, CRs are widely proposed to build DSAbased secondary networks for lower priority users.

One of the major functions of a CR is to find spectrum holes and be able to access and utilise them without causing any harmful interference to the incumbent or PU. A spectrum hole is defined as a band of frequency assigned to the PU, but which at a particular time and specific geographic location is not being used by that PU [11]. In the absence of signalling between PUs and SUs, spectrum holes may be identified by performing direct spectrum sensing, using geo-location databases, beaconing techniques, or by combining spectrum sensing with geo-location database information [7], [12]. (For interested readers the latest developments on database based CRNs, known as "SenseLess CRNs", are reported in [13].) A CR should also be intelligent enough to perform spectrum decision in order to select the most suitable frequency band to satisfy specific communication needs. Spectrum decision is a key function of CRs which requires greater attention in order to realize the practical implementation and deployment of CRNs. Consequently, the focus of this paper will be on spectrum decision frameworks in both centralized and distributed CRNs. Readers are referred to [2] and [11] for good introductions to spectrum agile CR technology.

## B. Cognitive Radio Network Topologies

A CRN is a wireless communication network whose enduser nodes are CRs. Similar to traditional wireless networks, a CRN topology can be classified as either centralized (infrastructure-based) or distributed (infrastructure-less or ad hoc) network topology. These network types are depicted in Fig. 3 and Fig. 4, respectively. In this paper, we consider both centralized and distributed CRN topologies.

# 1) Centralized CRN Topology

In the infrastructure-based CRN architecture, a central node such as a base station (BS) or access point (AP) is deployed with several SUs associated with it, as shown in Fig. 3.

A typical example of a centralized CRN is a IEEE 802.22 wireless regional area network (WRAN) or a cellular network. For simplicity, we shall take the IEEE 802.22 WRAN as an example to discuss a centralized CRN. However, similar reasoning can be applied to more complex centralized CRNs. In a centralized network, a BS controls all the SUs (clients) or consumer premises equipments (CPEs) within its transmission range. The CRN operates within the transmission or coverage area of the primary network. Thus it uses DSA techniques to opportunistically access the primary network spectrum without causing any harmful interference. To do this, all SUs perform spectrum observation on specified spectrum channels and then send their observations to the BS, which acts as a fusion centre. Both the BS and its associated clients may be capable of detecting the presence of the PUs using different detection techniques (such as spectrum sensing, geo-location databases or beaconing).

In some cases [14], two physical channels are used: one for observing the primary channel and the other for reporting data by the SUs to the BS. Once available channels are gathered, a BS will build the final list of these available channels and their associated maximum transmission powers, and then decide on the best channels to be accessed. These channels will then be broadcast back to all or selected SUs for use. In the next subsection we discuss the distributed CRN topology.

# 2) Distributed CRN Topology

In the distributed CR ad hoc network (CRAHN) topology, the SUs communicate directly with each other without any central or controlling node. As shown in Fig. 4, SUs share their local observations and analysis among themselves, as long as they are within each other's transmission range. For database-based networks, each SU may have access to query the database for available spectrum bands. Using both its results and the results of other SUs, a SU can make a decision for an appropriate band using a local criterion. If the criterion is not satisfied, the process may be repeated again until a decision is reached.

It is clear that spectrum decision in CRAHNs does not rely on a central node. However, if SUs decide to cooperate, as in cooperative spectrum sensing, one node can be chosen as the head node and be used for making spectrum decisions. Unlike in infrastructure-based topologies, spectrum decision in CRAHNs also involves route selection, which is normally addressed as a joint spectrum and route selection problem. A noticeable new challenge in CRAHNs, which did not exist in traditional wireless ad hoc networks, is that channel availability is determined by the present behaviour of PUs, which may vary with location, time and frequency [15]. Another new challenge is the re-routing and switching to other available channels or links once the PU appears on the occupied channel [16]. Thus the wide range of operating or available spectrum makes it infeasible to transmit beacons over all possible channels. Section V discusses these and many other spectrum selection challenges experienced in both distributed and centralized CRNs.

In this section, we have provided a brief overview of



Fig. 4: Distributed CRN Topology

CR technology in relation to spectrum decision and the two most commonly deployed CRN topologies (i.e. centralized and distributed topologies). In the next section, we present standardization and regulation efforts around CR technology.

#### **III. STANDARDIZATION AND REGULATORY EFFORTS**

The introduction of frequency agile CR technology created a spark in numerous academic, industry, regulatory and standardization bodies worldwide. Like any other new technology introduced to the market, CR technology's success will depend on sound standardization and regulation efforts from standardization bodies, regulators and industry. It is important to note that initial research and development efforts on CR technology have been focused in the United States. This is mainly due to the FCC's adoption of CR as an enabling technology for efficient spectrum management. Since then, other standardization bodies and regulatory agencies around the world have become interested in the standardization and regulation of CRs for DSA. In the following sub-sections we will discuss standardization and regulatory efforts on CRs, focusing mainly on efficient frequency management.

# A. Standardization

# 1) IEEE 802.22 Standard

The global switch-over from analogue to digital TV will leave a considerable amount of VHF/UHF spectrum vacant. The TVWS spectrum has excellent radio propagation characteristics, and is now being proposed as the most useful spectrum for improving wireless broadband connectivity in rural communities [12], [17]. In order to take advantage of the TVWS spectrum, the IEEE 802.22 WRAN standard [18] was established, and the first official standard was released in July 2011. IEEE 802.22 is the first wireless air interface standard focused on the development of CR based WRAN physical (PHY) and medium access control (MAC) layers for operation in TVWS. It specifies a fixed point-to-multipoint wireless air interface where a BS manages its own cell and all associated CPEs. The IEEE 802.22 PHY layer is based on orthogonal frequency division multiple access (OFDMA) and can support a system which uses TVWS channels to provide wireless communication links over distances of up to 100 km. A typical use case for the IEEE 802.22 standard would be in sparsely populated rural areas [7].

The IEEE 802.22 standard supports incumbent or PU detection through spectrum sensing techniques with an option for geo-location databases. However, there are still technical difficulties in performing reliable spectrum sensing practically. Thus, some regulatory bodies such as the FCC and Ofcom prefer geo-location databases as the primary means for incumbent detection. A beaconing option is also provided for incumbent user detection in IEEE 802.22. In IEEE 802.22, both the CPE and BS have the capability to detect the incumbent, but spectrum decision is only managed by the central BS. The BS employs the CR capabilities for spectrum decision based on the TV channels' operating characteristics [19]. The BS actually performs the spectrum characterization and selection functions, while the CPE is responsible for the reconfiguration of its transceiver parameters.

#### 2) IEEE DySPAN

After realizing the importance of coordinated work around CR standardization, the IEEE P1900 Standards Committee was jointly established by the IEEE Communications Society (ComSoc) and the IEEE Electromagnetic Compatibility Society in the first quarter of 2005 [20]. On 22 March 2007, the IEEE Standards Association Standards Board approved the reorganization of the IEEE P1900 activities as Standards Coordinating Committee 41 (SCC41), called Dynamic Spectrum Access Networks (DySPAN). The main aim of SCC41 is to develop supporting standards to address issues related to new technologies and the development of techniques for next generation radio systems and advanced spectrum management [21]. The SCC41 concentrates on developing architectural concepts and specifications for network management between incompatible wireless networks rather than specific mechanisms that can be added to the air interface.

In December 2010, the IEEE SCC41 was renamed the IEEE DySPAN-Standard Committee (DySPAN-SC). The IEEE DySPAN-SC consists of seven working groups (WGs), named 1900.1 through to 1900.7. Out of these WGs, the IEEE 1900.4's work has some elements of spectrum decision. This WG focuses on architectural building blocks enabling network-devices decision making for optimized radio resource usage in heterogeneous wireless access networks [20].

#### 3) European Telecommunications Standards Institute

In Europe, the European Telecommunications Standards Institute (ETSI) is also involved in the standardization of CR systems (called reconfigurable radio systems) under their Reconfigurable Radio Systems Technical Committee (RRS-TC) [22]. Cognitive radio principles within ETSI RRS-TC are concentrated on two topics: a cognitive pilot channel proposal and a functional architecture for management and control of reconfigurable radio systems. There are four WGs forming the ETSI RRS-TC, WG 1 to WG 4. Cognitive management and control falls under WG 3. This WG focuses on defining the system functionalities for reconfigurable and dynamic spectrum management and joint radio resource management. More information on ETSI RRS-TC can be found in [22].

## B. Regulation

The International Telecommunication Union (ITU) is also involved in standardization efforts of CR technology through their ITU-R Working Party (WP) 1B and WP 5A [23]. These two WPs prepared reports describing the concepts and the regulatory measures required to introduce CR. The ITU-R WP 1B developed a working document towards draft text on World Radio-communications Conference 2012 (WRC-12) agenda item 1.19. Agenda item 1.19 reads: "to consider regulatory measures and their relevance, in order to enable the introduction of software-defined radio and CR systems, based on the results of ITU-R studies, in accordance with Resolution 956 of WRC 07" [24]. The ITU-R WP 5A is currently developing the working document toward a preliminary new draft report, Cognitive Radio Systems in the Land Mobile Service [25]. This report will address the definition, description, and application of CR systems in the land mobile service [23]. The regulatory technicalities on dynamic spectrum management and spectrum decision from the ITU's point of view should become clearer after the WRC-12, where they will be discussed under agenda item 1.19.

Now that we have presented the necessary background to assist in understanding CR technology, CRN topologies, and CRN-related standardization and regulation activities, in the following sections we will discuss the three major functions in the spectrum decision framework. These will be followed by additional sections on spectrum decision in CR platforms and future developments in CR technology.

#### IV. SPECTRUM CHARACTERIZATION IN CRNs

In CRNs, multiple spectrum bands with different channel characteristics may be found to be available over a wide frequency range [26]. In order to properly determine the most suitable spectrum band, it is crucial to first identify the characteristics of each available spectrum band. Spectrum characterization allows the SUs to characterize the spectrum bands by considering the received signal strength, interference and the number of users currently residing in the spectrum, based on RF observation. The SUs should also observe heterogeneous spectrum availability which varies over time and space due to PU activities. Heterogeneous spectrum availability refers to the availability of spectrum holes which fluctuate over time and location and have different characteristics. Thus, spectrum characterization should include both the current RF environment conditions and the observed PU activity modelling. In this section, spectrum characterization in terms of the radio environment and PU activity models is discussed along with some related work. The section ends with key open research challenges in spectrum characterization.



Fig. 5: Radio frequency environment characterization elements

# A. Radio Frequency Environment Characterization

A CR is expected to continuously characterize radio environment usage in frequency, time and space. This is mainly due to the fact that available spectrum bands in CRNs always have different characteristics. Radio frequency (RF) environment characterization is a process that involves estimation of the following key elements or parameters: (1) channel identification, (2) channel capacity, (3) spectrum switching delay, (4) channel interference, (5) channel holding time (CHT), (6) channel error rate, (7) subscriber location, and (8) path loss [2], [27]. These elements are illustrated in Fig. 5 and analysed and discussed in the following subsections.

# 1) Channel Identification

Primary channel identification is the first important step to be performed by each CRN. As research in CR evolves, different CRNs application areas are being introduced to the market. Some of these applications include: television (TV) white space networks, smart grid networks, machine-tomachine (M2M) networks, public safety networks, broadband cellular networks and wireless medical networks [28]. These applications or networks exhibit different traffic data patterns, either deterministic or stochastic.

In deterministic traffic, the PU is assigned a fixed time slot on a frequency band for communication. Once the PU stops communicating, the frequency band becomes available and can be used by the SUs. Examples of deterministic traffic data patterns occur in TV broadcasting (longer periods) and radar transmitters (shorter periods). Normally deterministic signals have fixed or predictable ON and OFF periods which can be determined by a mathematical expression, rule or table. Any future value for a deterministic signal can be calculated or predicted based on its past values, which makes it easy to predict future PU idle periods for CRNs.

On the other hand, stochastic traffic patterns can only be described and analysed using probabilities and statistics because their spectrum usage tends to exhibit greater variations in time and space [11]. Due to their randomness, stochastic signals are analysed using average values from a collection of primary signals. Examples of stochastic traffic data patterns occur in cellular networks. In order to improve the accuracy of stochastic traffic modelling, Haykin [11] suggested the design of a tracking strategy for PU idle period prediction models.

A prediction method for both deterministic and stochastic traffic patterns is proposed in [29]. This method is used by CRs to predict the primary channel idle time. In summary, channel identification is a crucial function for every CRN for learning its external environment, classifying primary traffic and applying the appropriate spectrum decision methods.

# 2) Channel Capacity Estimation

While many other parameters (such as the channel interference level, error rate, path loss, delay and holding time) are important for efficient spectrum characterization, a considerable amount of research has focused on channel capacity estimation. This is motivated by the fact that by estimating other (above) channel parameters, we can determine the channel capacity [2] (this means channel capacity can be derived from the above parameters). It has been shown that the traditional method of estimating channel capacity using the signal-to-noise-ratio (SNR) leads to non-optimal spectrum decision [30].

In an orthogonal frequency division multiplex (OFDM) system, each spectrum band *i* has a different bandwidth  $B_i$ , consisting of multiple subcarriers. A normalized CR capacity  $C_i^{CR}(k)$  model of spectrum band *i* for user *k* is proposed in [26] for spectrum characterization in CRNs. This  $C_i^{CR}$  model defines the expected normalized capacity of user *k* in spectrum band *i* as:

$$C_i^{CR}(k) = E[C_i(k)] = \frac{T_i^{off}}{T_i^{off} + \tau} \cdot \gamma_i \cdot c_i(k)$$
(1)

where  $C_i(k)$  represents the spectrum capacity, the term  $c_i(k)$  (with small c) is the normalized channel capacity of spectrum band i in bits/sec/Hz,  $\tau$  represents the spectrum switching delay,  $\gamma_i$  represents the spectrum sensing efficiency and  $T_i^{off}$  is the expected transmission time without switching in spectrum band i. Spectrum or channel switching delay is introduced within CRNs when SUs move from one spectrum band to another according to PU activity. More on channel switching delay is discussed in the next sub-section. Spectrum sensing efficiency arises due to the fact that RF front-ends cannot perform sensing and transmission at the same time, which inevitably decreases their transmission opportunities. When using spectrum sensing to detect spectrum holes, sensing efficiency is influenced by the observation time and transmission time [31].

#### 3) Channel Switching Delay

In opportunistic or DSA network settings, radio nodes are expected to operate on different frequency channels in a given time without disrupting existing network connections [32]. This process is commonly known as dynamic channel switching (DCS). In CRNs, channel switching may be triggered by the detection of PUs on the operating channel, by degradation of the QoS due to interference, or traffic load in the current channel [18]. During the channel switching process, the SUs must dynamically switch from one channel to another idle channel, and during this switching process all SUs transmissions are temporarily suspended until the new spectrum opportunity or channel is found. Switching from one channel to another introduces additional delay to the CRNs, which is called switching delay. This switching delay may vary from one node to another since it depends on the hardware technology (e.g. time spent during RF front-end reconfiguration) and the algorithm used by the SUs or the overall CRNs to perform the spectrum decision process. The switching delay may also include the sensing time in cases where available channels are detected using spectrum sensing techniques. The ultimate goal is to keep the switching delay as short as possible to ensure that it does not affects the overall CRN performance.

Azarfar et al. [33] addresses the channel switching delay as an overhead that has the potential to decrease the useful time available for data communication. They consider channel switching delay as the sum of spectrum sensing duration plus the channel recovery time. Channel recovery time is the time spent by a SU to vacate the channel, to decide on the available channel (or signalling time for establishing new channels), and to select or access the available spectrum. Several techniques are proposed in [33] to reduce the channel switching delay. This includes the use of historical information of channel occupancy and channel quality index [33]. Historical information of channel occupancy becomes useful provided the SUs are aware of their operating location. For instance, a SU may use geographical coordinates of the previously explored areas to remember that spectrum band X was occupied or vacant and also to remember which technologies operate in that area. So when the SU approaches those areas (i.e. previously explored areas), it can save time by avoiding those spectrum bands which were found to be occupied during the previous visits. This will then reduce the channel recovery time, thereby reducing the channel switching delay.

In centralized CRNs such as IEEE 802.22 [18], the BS can decides to switch channels during normal operations by first selecting the backup channel from the backup or candidate channel list. Secondly, it must wait for a specified time to make sure that all the associated CPEs are prepared for the channel switch. This waiting time should be long enough for the CPEs to recover from an incumbent detection. Once all CPEs are prepared for channel switching, the BS can then schedule the channel switching procedure. Typical spectrum switching delay in IEEE 802.22 is less than 2 seconds [18].

Xu *et al.* [34] analysed a trade-off between higher bandwidth and switching overhead experienced in CRNs. In their paper, switching overhead is the sum of spectrum sensing time, channel evacuation time, and link setup time. The link setup and channel evacuation times are based on the radio hardware and the operating environment [34], and are modelled as a random variable for all SUs. It was found that using higher bandwidth in saturated traffic (i.e. where there is a large pool of idle PU channels to be sensed) with few SUs leads to more switching overhead due to an increased sensing time. However, in cases where a central node provides channel availability information to multiple SUs (i.e. SUs do not need to conduct spectrum sensing), the switching overhead reduces (it only consists of link setup and channel evacuation times).

In distributed CRNs, channel switching overhead on a node includes the switching delay, other flows' transmissions delay and the back-off delay (i.e. interference within a frequency channel) [35]. A delay based routing metric called cumulative delay for on-demand routing protocol is proposed in [35]. This cumulative delay is the total path delay derived from the spectrum switching delay and back-off delay. The backoff delay arises when more than two nodes contend for the spectrum resource. This delay will depend on the number of contending nodes on each spectrum band. Thus during spectrum characterization, the number of SUs in a distributed CRNs is important to estimate the switching delay.

An analysis of delay performance in CR sensor networks is presented in [36]. Two types of channel switching techniques are considered: periodic switching and triggered switching. In periodic switching, SUs switch to a new channel only at the beginning of each channel switching interval as defined in [36]. Periodic switching occurs if the occupied channel becomes unavailable (to the SUs) before the end of the channel switching interval. Channel switching can be caused by the sudden appearance of PUs or high interference on the channel due to adjacent channels. In triggered switching, the SUs switch to a new channel as soon as the current channel is no longer available [36]. It was found that bursty traffic experiences shorter delays when considering periodic switching techniques as compared to triggered switching techniques. This was mainly due to the fact that a channel is likely to be available in the earlier portion of the channel switching interval than in the later portion.

# 4) Channel Interference Estimation

In CRNs environment, SUs are expected to coexist with licensed or primary users (PUs). In some cases, several CRNs must also coexist within the coverage area of single or multiple primary networks. Such coexistence, if not controlled, can lead to harmful interference to the PUs. It is therefore crucial to accurately estimate and model interference generated by multiple active SUs in the network. In this subsection, different techniques for estimating, controlling and modelling channel interference caused by CRNs are discussed.

In [37], an opportunistic interference alignment scheme that allows multiple SUs to exploit the unused spatial dimensions of multiple-input-multiple-output (MIMO) PU channels is proposed. In this scheme, the primary transmitters maximize their rate by water-filling over the singular values of their channel matrix. By using singular values, PUs leave some eigen-modes unused, which allows the SUs to transmit at a significant rate by aligning their signals along the free eigenmodes of the PU channel. As a result, this scheme protects the transmission of the PUs while providing interference-free communication for the SUs.

An interference-aware radio resource allocation scheme is proposed in [38]. The paper studies PU interference caused as a result of: CR out-of-band (OOB) emissions and the interference that arises as a result of imperfect spectrum sensing. In OFDM-based primary networks, OOB emissions are due to power leakage in the side-lobes of transmitted signal. The amount of OOB interference power introduced in a PU sub-carrier due to SU transmission is modelled for both the uplink and downlink sub-bands. Finally, a computationally efficient algorithm for downlink and uplink subcarriers and power allocation in an OFDMA-based CRN is developed based on proposed OOB emissions and imperfect-spectrumsensing-based interference models. To minimize the amount of OOB interference generated via non-contiguous multicarrier data transmission, multi-rate filter banks are suggested in [39]. In this approach, the multi-rate filter banks' sub-band spectra can be designed to be highly spectrally selective to limit the amount of intercarrier interference (ICI), which becomes advantageous in cellular systems that suffer from Doppler effects and frequency selectivity of the channels.

In [40], two interference detection schemes are proposed. The first scheme is based on pilot-aided interference detection for OFDM systems. In order to detect the presence of interference, this scheme requires at least two pilot symbols in a given subcarrier spaced in time. The pilot symbols are designed to ensure that their summation or subtraction is zero. For instance, two pilot symbols,  $x_1^p$  and  $x_2^p$ , can be selected as the two points in the Binary Phase Shift Keying (BPSK) constellation such that their summation is zero (i.e.  $x_1^p + x_2^p = 0$ ). Although this pilot-aided scheme is simple in implementation, its weakness is poor interference detection in the sub-carriers where no pilot exists due to sparse placement of the pilot symbols. The second scheme proposed in [40] is based on a joint interference detection and decoding technique which does not require any pilot symbols. The decoder jointly performs erasure marking and decoding in order to erase the interference jammed symbols automatically during the decoding process. The decoding process consist of two steps: the first step determines the positions of the erasures, and the second step determines the number of erasures. However, this technique suffers from increased computational complexity and decoding delay.

Rabbachin *et al.* [41] proposed a statistical model for perdimension (real or imaginary part) aggregate interference of a CRN which allows modelling of the CRN interference generated by SUs in a limited or finite region. In this model, two types of secondary spatial reuse protocols are considered: single-threshold and multiple-threshold protocols. For each protocol, the characteristic function of the CRN interference is expressed, and then used to derive its cumulants. The cumulants are used to model the CRN interference as truncatedstable random variables. The interference signal at the primary receiver generated by the  $i^{th}$  cognitive interference is modelled as:

$$I_i = \sqrt{P_I} R_i^{-b} X_i \tag{2}$$

where  $P_I$  is the interference signal power at the limit of the near-far region (which is limited to 1m),  $R_i$  is the distance between the  $i^{th}$  cognitive interferer and the primary receiver, b is the amplitude path-loss exponent, and  $X_i$  is the per-dimension fading channel path gain of the channel from the  $i^{th}$  cognitive interferer to the primary receiver.

#### 5) Channel Holding Time

Channel holding time (CHT) is the expected duration the SUs can occupy a licensed band before getting interrupted. Thus, the longer the holding time, the better the QoS for the SUs [2]. CHT can be determined by the type of secondary services served by the CRN or it can be determined by the regulator. It is useful in determining the RF environment characterization.

In [42], a Markovian model for finding the duration of the spectrum hole is proposed. This model builds on a CHT concept for the PU. Once the PU idle time is modelled as CHT, matrix-analytic techniques are applied to derive and analyse the duration of the spectrum holes which can be accessed by the SUs. One of the main drawbacks of this technique lies in its complexity.

Yuan et al. [43] introduced the concept of time-spectrum block to model spectrum reservation for CRNs. A timespectrum block concept represents the time for which a SU occupies a portion of vacant spectrum without causing interference to the PUs. For a CRN of n nodes  $(V = v_1, \dots, v_n)$ , located in the two-dimensional Euclidean plane, let  $d(v_i, v_j)$  denote the Euclidean distance between  $v_i$  and  $v_j$ . A time-spectrum block  $B_{ij}^k = (t_k, \triangle t_k, f_k, \triangle f_k)$ is assigned to link  $(v_i, v_j)$  if sender  $v_i$  is assigned the contiguous frequency band  $[f_k, f_k + riangle f_k]$  of bandwidth  $\triangle f_k$  during time interval  $[t_k, t_k + \triangle t_k]$ . Using the above definitions, the dynamic spectrum allocation problem can be viewed as dynamic packing of time-spectrum blocks into a three-dimensional resource, consisting of time, frequency, and space. Using a time-spectrum block  $B_{ij}^k$ , it is possible to find the time overhead of switching frequency or the time used for medium access contention in CRNs.

#### 6) Channel Error Rate

In a communication link, error rate is defined as the rate of bits or data elements which are incorrectly received from the total number of bits or data elements sent during a specified time interval [44]. The average channel error rate is a useful parameter in estimating the RF environment characterization in CRNs. It depends on the interference level (interference to SU may be caused by the primary transmitters or other SUs), the available bandwidth, the frequency band in use and the modulation scheme (or access technology) [2]. Bit error rate (BER) and frame error rate (FER) are the most commonly used metrics. Error rate is usually stated relative to the channel's signal to noise ratio (SNR) values, and this makes the transmitted energy per bit an important metric in error estimation [44].

A closed-form average BER expression is derived in [45] to investigate SU's error performance for the binary phase shift keying (BPSK) modulation scheme. It was found that the channel error performance improves when the SU's SNR increases. Kaur and Sharma [46] analysed the bit error rate (BER) performance of the CR PHY layer over Rayleigh fading channels under different channel encoding schemes and channel conditions. They considered CRNs with both contention-based non-persistent carrier sense multiple access

# (CSMA) and OFDMA techniques.

#### 7) Subscriber Location

Subscriber location also need to be determined when characterizing the RF environment. Generally, a SU can obtain geographical and environmental information using built-in global positioning system (GPS) coordinates, embedded information in packets exchanged between nodes or a central server that sends the most up-to-date global Radio Environment Map (REM) information [33]. IEEE 802.22 defines two modes which the SUs and BSs can use to find their geo-location: satellite-based geo-location (which is mandatory) and terrestrial-based geo-location (which is assisted by the CDMA ranging) [47]. By knowing its location, a SU can record a number of normal and abnormal events experienced in different locations and times. Such knowledge will be useful for future predictions of spectrum holes and characterization of the RF environment.

#### 8) Path Loss

Path loss, the deterministic overall reduction of received signal power with distance between the transmitter and the receiver, is one of the factors affecting the radio propagation across wireless channels [11]. It is caused by the spreading of the electromagnetic wave radiating from the transmit antenna and the obstructive effects of the surrounding objects [44]. Path loss is normally obtained at the receiver side by dividing the transmitted power with the received power. Thus, transmission power can be increased to compensate for the increased path loss. However, this might cause higher interference for other SUs and PUs. According to Rappaport [48], the average path loss of a channel can be expressed using a path loss exponent ( $\alpha$ ). This path loss exponent  $\alpha$  depends on frequency, antenna heights, and propagation environment. In free space loss, the path loss exponent  $\alpha$  is equal to 2.

In [49], a low-complexity adaptive transmission protocol for CRNs, whose links have unknown and time-varying propagation losses, is proposed. This protocol adjusts its transmission modulation and coding as a mechanism for responding to changes in propagation losses. The transmitter power is increased only if the most powerful combination of coding and modulation is inadequate.

In this subsection we reviewed related work on RF environment characterization, which is the first step towards reliable spectrum decision scheme development. In the next subsection, we discuss PU activity modelling as another aspect of spectrum characterization.

# B. Primary User Activity Modelling

Because there is no guarantee that a spectrum band will be available during the entire SU communication period, it is important to consider how often the PUs appear on the spectrum band. Using the learning ability of the CR, the history of the spectrum usage information can be used for predicting the future profile of the spectrum. This process is achieved through *PU activity modelling*. By considering the PU activity, the SUs can decide on the best available spectrum bands to be used for their transmissions. For a stable CRN, the spectrum decision framework must be aware of the available spectrum fluctuations, as well as the heterogeneous QoS requirements of the SUs [2]. It is important for the spectrum decision function to also consider the spectrum fluctuations because the SUs can transmit data only if they can accurately detect the spectrum holes. PU activity modelling also plays a crucial role in the design of communication protocols for CRNs [50]. For example: once we know that a PU favours a particular channel and tends to occupy it for a long period of time, that channel would be less likely to be available for a SU, as a result, sensing on such a channel would likely be a waste of time and energy [51]. Therefore, PU activity modelling will result in more effective spectrum usage for SUs, which in turn enhances CRNs performance [52]. Different techniques are used to model PU activities, and we discuss them in the next sub-sections.

#### 1) PU Activity based on Poisson Modelling

There is a significant amount of research that models the PU activity as a Poisson process with exponentially distributed inter-arrivals [26], [53], [54]. The PU traffic is modelled as a two-state birth-death process with death rate  $\alpha$  and birth rate  $\beta$ . In this approach, each user arrival is independent, and the PU transmission is assumed to follow the Poisson arrival process. As a result, the length of ON and OFF periods are exponentially distributed. An ON state represents the period used by PUs and an OFF state represents the vacant periods. An adaptive spectrum decision framework is proposed in [26] to determine a set of spectrum bands to satisfy the user requirements under the dynamic nature of RF spectrum bands. In this framework, each spectrum band is first characterized by jointly considering PU activity and spectrum sensing results.

For real-time applications, a minimum variance-based spectrum decision which minimizes the capacity variance of the decided spectrum bands is proposed. For best-effort applications, a maximum capacity-based spectrum decision scheme is proposed to maximize the total network capacity. However, based on a large-scale measurement-driven characterization of the PU activity in cellular networks investigated in [55], it was found that PU activity durations are non-exponential and high fluctuations *of the PU activity* may violate the Poisson assumption.

While the majority of studies assume that the PU activity follows the Poisson model, such assumptions were invalidated in [52], [56] based on the following reasons:

- The Poisson model approximates the PU activities as smooth and burst-free traffic. Therefore it fails to capture the bursty and spiky characteristics of the monitored data.
- The Poisson model does not consider correlations and similarities within data.
- The Poisson model fails to capture the short-term temporal fluctuations or variations exhibited by the PU activity.

Therefore the existing research which assumes the Poisson model derives PU activity models that have smooth and burstfree traffic in which short-term fluctuations are neglected [52]. A good illustration of the Poisson model's drawbacks can be found in [52], as depicted in Fig. 6. The Poisson model



Fig. 6: Missed Opportunities in Poisson Model [52]

represents the ON period (which is the active transmission duration of a PU) and the OFF period (which represents the absence of PU activity). It can be observed from Fig. 6 that the actual PU activity fluctuates during the ON period, which is not tracked by the Poisson model. As a result, these durations are classified as part of the ON period, which leads to missed transmission opportunities. Based on these reasons, it is clear that Poisson modelling may lead to performance degradation in CRNs due to unidentified fluctuations in PU activities. In order to address the potential Poisson model drawbacks, the following subsections presents alternative mechanisms for PU activity modelling.

# 2) PU Modelling Based on Statistics

A simple method for learning and classifying traffic patterns on primary channels is proposed in [29]. The authors consider a primary network that consists of multiple channels with independent traffic patterns. The method starts by collecting spectrum usage information (through spectrum sensing) and stores this information in the channel database (in binary format). If the channel is free, the channel state (CS) flag is set to 0, and CS is set to 1 if the channel is occupied. Once the spectrum information is stored, the traffic patterns of each channel are classified as either stochastic or deterministic. The channel classification algorithm used is based on the periodicity, where an edge detection method is used for period search. The average separation of the raising edges and the standard deviation of the separations (of the edge) are used to determine whether the traffic is deterministic or stochastic. After classifying traffic type, the idle time prediction method is selected. To predict the PU idle time, the CS information (stored in the channel database) is used. If CS = 0, the idle time prediction is set to 1. If CS = 1, the idle time prediction is set to 0. Secondary transmission will continue on channels with longer idle time, and once the PU appears on this channel, the SU will switch to the channel with the longest expected remaining idle time. Fig 7 summarizes the model as proposed



Fig. 7: Prediction System Model [29]

in [29].

Acharya *et al.* [57] proposed a predictive model based on long-term and short-term usage statistics of TV channels. In this model, the usability characteristics of a channel are based on TV channel statistics and are later used for selection of a channel for opportunistic transmission. The model uses a threshold mechanism to filter out channels with frequent and heavy appearance of PUs.

In [58], a multivariate time series approach is used to learn the PU characteristics and then predict the future occupancy of neighbouring channels. In order to reduce the complexity and storage requirements, a binary scheme is used; where 0 represents empty and 1 represents occupied channels.

# 3) PU Modelling Based on Measured Data

Riihijarvi *et al.* [59] proposed a technique to characterize and model the spectrum for DSA with spatial statistics and random fields using measured data. The proposed techniques were implemented using data obtained through spectrum measurements. First, they perform spectrum characterization by treating spectrum measurement results as a realization of some unknown random field Z. In this case, random fields are considered as extensions of the theory of stochastic processes from one-dimensional time to multi-dimensional space. Different metrics suitable for characterizing second-order stochastic spatial structure of the primary spectrum are derived using *Moran's I* [60] and *Geary's C* [61] statistics. These are two statistics commonly used to measure the degree of spatial correlation for spatial data whose covariance structure is defined by neighbourhoods.

Moran's I statistics is defined as:

$$I = \frac{n}{S_0} \cdot \frac{\sum \sum W_{ij} (Z_i - \bar{Z}) (Z_j - \bar{Z})}{\sum (Z_i - \bar{Z})^2}$$
(3)

where *n* is the sample size,  $W_{ij}$  is a matrix of weights (where  $W_{ij} = 1$  if site i, j are neighbours, otherwise  $W_{ij} = 0$ ).  $S_0 = \sum \sum W_{ij} =$  twice the number of neighbours,  $Z_i$  indicates the continuous response at site *i*, and finally  $\overline{Z}$  denotes the usual estimate for the mean of *Z*. The value of Moran's I lies between -1 and +1, where former indicating strong negative autocorrelation and latter strong positive autocorrelation. See [59] for Geary's C statistics representation. The above spatial statistics models were tested on distributed spectrum occupancy measurements of frequency bands used by popular wireless services such as Wi-Fi, TV and cellular networks. However, it was found that Geary's C requires sufficient amount of data to ensure that existing similarity patterns are not missed when the investigated bands are only rarely (i.e. in less busy locations/times).

A time-varying statistical model for spectrum occupancy using actual wireless frequency measurements is proposed in [62]. Using statistical characteristics extracted from actual RF measurements, first-order and second-order parameters are employed in a statistical spectrum occupancy model based on a combination of different probability density functions.

# 4) Other PU Modelling Techniques

Canberk *et al.* [52] developed a real-time based PU activity model for CRNs using first-difference filter clustering and correlation. In this difference model, the PU signal samples are first collected over a pre-determined duration, then the observed PU signals are clustered together if they are greater than a threshold. The authors developed a *PU activity monitoring module* which is implemented within each SU to monitor the spectrum bands and then samples the PU activity. In simple terms, this module is responsible for local spectrum sensing by testing the well known  $H_1$  and  $H_0$  hypotheses [2], where  $H_1$  indicates that the PU has an activity in the spectrum band and  $H_0$  indicates that there is no PU activity in the spectrum band. This PU activity monitoring module stores samples of the monitored PU activities into a vector, q, of size p as [52]:

$$q = [q(1), q(2), \cdots, q(m), \cdots, q(p)] \in \Re^p,$$
 (4)

where m is the sampling index and p is the total number of PU monitored activity samples. This model claims to produce accurate PU estimation and higher throughput than the Poisson model.

# C. Open Research Issues in Spectrum Characterization

#### 1. Secondary user activity modelling

Most of the available literature focuses on the modelling of PU activities and neglects the modelling of SU behaviour. With the anticipated growth in the number of CRNs, it is important to devise reliable models for SU behaviour and characteristics.

# 2. Heterogeneous Users Activity Database

Spectrum characterization involves reliable modelling of PU activities. It has been suggested in [14] that a database can be used to store all the knowledge of the PU radio frequency environments. A major challenge is to create a reliable and secure database to store both the PUs and SUs activity models.

In summary, spectrum characterization allows CRNs to be aware of their operating RF environment and to intelligently determine the ongoing PU activities in a licensed spectrum. Using the learning ability of the CR, the history of the spectrum usage information can be used for predicting the future profile of the spectrum. In this section, we have discussed the key issues in RF environment characterization and PU activity modelling. Table I summarises the PU activity modelling techniques (with advantages and disadvantages) discussed in this paper. We have also discussed some open research issues for spectrum characterization in CRNs. In the next section, we discuss spectrum selection as another key function of spectrum decision.

TABLE I: Primary User Activity Modelling: Summary

Modelling Tech-	Advantages	Disadvantages
nique		
Poisson	Widely used and easy to	Fails to capture
Modelling	model traffic	short-term temporal
		variations; does not
		consider correlations
		and similarities within
		data
Statistics	Predicts future PU idle	High probability of SU
Modelling	time based on history	collision with PUs
	and learning	
Measured Data	Uses real measured data	Low computational
Modelling		complexity

# V. SPECTRUM SELECTION IN CRNs

Once spectrum holes are characterized, the next major step is to select the best available spectrum suitable for the user's specific QoS requirements. In CRAHNs, the set of channels available for each node is not static. Due to dynamically changing topologies and varying RF propagation characteristics, spectrum selection techniques in CRAHNs should be closely coupled with routing protocols (commonly called joint route and spectrum selection). In this section, we discuss related work on addressing the spectrum selection problem in centralized and distributed CRNs. We conclude the section with some key open research challenges.

# A. Spectrum Selection in Centralized CRNs

A typical centralized CRN is the IEEE 802.22 WRAN standard, which specifies a fixed point-to-multipoint topology. The cognitive function of IEEE 802.22 is in dynamic channel management [19]. Spectrum availability in IEEE 802.22 depends on whether a TV channel is occupied by incumbent users or not. Each TV channel occupies a frequency bandwidth of 6 to 8 MHz, depending on the country of operation. For example, in South Africa, each TV channel occupies a bandwidth of 8 MHz. Besides TV services, wireless microphones are also incumbent users on TV channels. These wireless microphones occupy very narrow bandwidths (in the range of 200 kHz) of the TV channel. It is important for a CRN operating on TVWS to support variable channel widths when selecting specific channels for different services or applications. The advantages of variable channel widths are discussed in [63], where it is shown that decreasing the channel width increases the communication range. This is mainly due to improved SNR (i.e. higher SNR for narrower widths) and resilience to delay spread. The delay spread resilience is shown in [63] for an OFDM network, "where the guard interval increases by a factor of two each time the channel width is halved". The mixture of TV and wireless microphone users within the TVWS makes it important for a BS to also consider

narrow bandwidth incumbent users when performing spectrum selection.

Spectrum selection in centralized networks normally lies at the BS or AP. In TVWS based networks (such as IEEE 802.22), the biggest spectrum selection challenge is the fragmentation of the available frequency. This fragmentation varies from one channel to several channels, depending on the density of the TV stations. In [64], a spectrum assignment algorithm for managing variable channel width is proposed. This algorithm, called signal interpretation before Fourier transform, uses SDR to perform time-domain analysis of the raw signal in order to determine the available channel width. A moving average over a sliding window of the signal amplitude value is computed for accurate detection of the beginning and end of packet transmission.

In [65], a non-cooperative game theoretic framework is proposed to evaluate spectrum management functionalities in CRNs, with a particular focus on spectrum selection. In this framework, the spectrum selection process is modelled as a non-cooperative game among SUs who can opportunistically select the best spectrum opportunity. Two different cost functions that include the number of interferers, the bandwidth and the expected holding time are considered. The first cost function is defined as a linear combination of the three factors: interference, bandwidth and holding time. The second cost function considers the product of these three defined parameters.

A policy-based spectrum selection architecture for centralized CRNs is proposed in [66]. Three different pre-defined policies for spectrum selection are presented: weighted selection, sequential selection, and combined selection. In weighted selection, weights are given to each selection criterion within the BS and the best channels are selected based on the sum of weighted values. In sequential selection method, the policy manager provides the channel selection order and the channel selection procedure continues until there are no more channels (i.e. identified vacant channels) left. And finally, the combined selection method uses both the weighted and sequential methods. Several criteria are used for the BS to make the proper spectrum or channel selection. This includes coverage area, number of supported nodes, average SNR, and channel occupancy history.

# B. Spectrum Selection in Distributed CRNs

In distributed multi-hop CRNs, such as CRAHNs, the entire communication session consists of multiple hops with rapidly changing channel properties and switching from one channel to another (heterogeneous spectrum availability). This imposes new challenges for designing optimal routing protocols. In order to address spectrum selection in multi-hop CRNs, a joint spectrum and route selection design approach is preferred [15], [16], [35], [67]–[71]. In joint spectrum and route selection design, lower layer (such as MAC layer) knowledge of the wireless medium is shared with the network layer [16] in order to make an intelligent decision. In this subsection we review related joint spectrum and route selection in multi-hop CRNs.

Routing in traditional multi-hop wireless networks is a well known and well studied problem. The emergence of multi-hop CRNs brings new challenges to routing which call for new cognitive routing approaches with novel metrics that capture spectrum availability and PU activities [68]. If we decouple the route and spectrum selection, each of these tasks will be distributed to the network and MAC layers, respectively. Although it offers an apparently simple solution to managing spectrum in multi-hop networks, decoupling route and spectrum selection suffers from poor prediction of link quality and also fails to address end-to-end optimization, which is important for multi-hop transmission [69]. A comparative study between two design methods, (1) decoupled route and spectrum selection, and (2) joint route and spectrum selection, is presented in [69]. The authors note that joint route and spectrum selection design allows each SU node to select the packet route, the channel to be used by each link on the route, and a time schedule of the channel usage [69]. In this paper we focus more on the joint route and spectrum selection method due to its relevance to our theme, and readers are referred to [69] for the decoupled route and spectrum selection method.

A joint routing and spectrum selection protocol which computes paths from a source to a destination by considering the PU activities and the path availability is proposed in [68]. This protocol exploits the parallel transmission capability of CR nodes and calculates the link capacity on every channel dynamically. In [70], a probability based routing metric is presented. This metric relies on the probability distribution (which is assumed to be a log-normal distribution) of the PUs to SUs interference at the SU node over a given channel. The path selection algorithm is initiated by the source node whenever an application requests a route to a destination. The source node acquires information about other nodes through link state advertisements over the control channel.

Ju and Evans [16] introduced a scalable cognitive routing protocol (SCRP) for mobile ad-hoc networks which employs an intelligent flooding protocol to save on routing overhead. This is achieved by allowing nodes to selectively flood route request (RREQ) packets along predicted strong links and over predicted preferred frequencies or channels. Unlike traditional on-demand routing protocols, SCRP makes use of neural network machine learning methods to make the CR nodes aware of history.

One of the challenges in joint spectrum and route selection is re-routing and switching to other available channels or links once the PU appears. To address the problem of SU link disconnections, Shih et al. [15] proposed a route robustness approach for path selection in multi-hop CRNs. This is done by guaranteeing a basic level of robustness for a set of routes, and then selecting some routes from this set and determining the spectrum to be allocated on each link. In [35], a joint interaction between on-demand routing and spectrum scheduling is proposed where an analytical model is used to describe the scheduling-based channel assignment progress. This model reduces the inter-flow interference and frequent channel switching delay. A distributed algorithm that jointly solves the routing, spectrum assignment, scheduling and power allocation in multi-hop CRNs is proposed in [71]. Through this method, each SU node makes real-time decisions on spectrum and power allocation based on locally collected information. Thus, SU nodes can adjust their transmission power, in order to maximize link capacity, based on the assigned spectrum band.

#### C. Open Research Issues in Spectrum Selection

#### **1.** Cooperative Spectrum Selection

Cooperation has been at the forefront of research in CRNs due to its advantages over non-cooperative approaches. A good example is in cooperative spectrum sensing, where neighbouring SUs share their sensing information with the aim of exploiting spatial diversity. A challenge in cooperative spectrum selection is on how to combine information from cooperating users while addressing the transmission or cooperation overhead.

# 2. Spectrum Selection in Heterogeneous Traffic Networks

In a given CRN, the SUs may have heterogeneous QoS requirements. The available spectrum may exhibit fluctuating and variable spectrum qualities. In heterogeneous traffic networks, a challenge is to select appropriate bands to satisfy the heterogeneous QoS requirements of each SU.

#### 3. Frequency Switching Delay along Multiple Hops

In multi-hop CRNs, each intermediate SU node receives packets on one frequency channel, switches its transceiver to a different frequency channel, and then transmits the packets to the next node. The time a packet takes to reach its destination, after traversing multiple nodes, results in a cumulative channel switching delay. There is therefore a need to develop cognitive routing protocols which minimize the cumulative delay along the entire path.

In this section, we have discussed the importance of spectrum selection in CRNs for distributed and centralized network topologies. In centralized topologies, spectrum selection decision is made by the central node such as the BS or AP. The situation is more complex in multi-hop CRNs, which calls for a joint spectrum and route selection approach. Although joint spectrum and route selection design offers improved accuracy and reliability, it comes with additional complexity and communication overhead. Table II briefly summarizes selection techniques discussed in this paper. We concluded the section by listing some open research issues for spectrum selection in CRNs. For additional open research issues in multi-hop CRNs, readers are referred to [72].

#### VI. RECONFIGURATION IN CRNs

In traditional wireless networks, the radio terminals are statically configured to operate over pre-defined frequency channels with pre-defined transceiver parameters and characteristics. Although such systems may employ adaptive techniques to adjust various transmission parameters such as transmission power, and modulation and coding schemes, their hardware-based architecture limits their flexibility to adapt to the external environment. However, heterogeneous spectrum availability and DSA requires systems that are far more flexible. CRs offer such flexibility and are able to rapidly adapt their transceiver parameters (such as channel width,

TABLE II: Spectrum Selection Solutions: Summary

Selection Tech-	Advantages	Disadvantages
nique		
Centralized	Controlled at the BS or	Single point of fail-
selection (e.g.	AP and standard based.	ure and fragmentation of
IEEE 802.22)		channels.
Joint routing	Efficient spectrum us-	Computational
and spectrum	age; good end-to-end	complexity;
selection	performance.	communication
		overhead; and re-
		routing due to channel
		switching.
Decoupled rout-	Simple to implement	Unreliable route selec-
ing and spectrum	since it uses existing	tion due to spectrum
selection	solutions.	fluctuation; poor end-to-
		end performance; and
		requires multi radios



Fig. 8: A CR Reconfiguration Manager

centre frequency, transmission power, modulation and coding) based on the external RF environment, policy updates, QoS requirements, selected spectrum, channel characteristics and the needs of the users. This is illustrated in Fig. 8. This flexibility is easily realised by implementing CRs using SDRs [27].

The CR reconfiguration task requires a clear understanding of how the communication parameters interact within the different protocol layers. Based on our spectrum decision framework (see Fig. 2), reconfiguration of parameters occurs after the spectrum of choice has been characterized and selected. In this section, we first discuss reconfigurable parameters in a CR, and then present some related work on CR reconfiguration in both centralized and distributed CRNs. We also discuss energy efficiency in CRNs because it affects the reconfigurable radio parameters. We conclude this section with open research issues.

# A. Reconfigurable Parameters

In this subsection, we study radio parameters that are commonly reconfigured in CRNs in order to adapt and satisfy the QoS requirements and regulatory policies. We limit our discussion to the parameters depicted in the reconfiguration manager in Fig. 8.

## 1) Modulation and Coding Schemes

A CR should reconfigure the modulation and coding schemes to adapt to changing user requirements and channel conditions [2]. An adaptive transmission scheme based on a simplified transmission scenario and environment for CRNs is proposed in [73]. Based on the interference temperature model, the proposed scheme adaptively selects the modulation order that provides the maximum throughput for the SUs in the given channels.

# 2) Transmission Power

Power control is used to manage and adjust the transmitted power of wireless nodes in order to achieve objectives such as reducing co-channel interference, managing data quality, maximizing network capacity and minimizing energy usage [74]. An effective CR power control scheme must be a tradeoff between the protection of PUs from harmful interference and the support of SUs' QoS, as long as the interference constraint is not violated [75]. Unlike conventional radios, a CR adapts its transmission parameters based on the environment that it operates in. The transmission power of a SU affects the communication range and also how often the SU sees spectrum opportunities. If a SU wants to transmit at high power levels, it must wait for the opportunity to do so (meaning it should wait until there is an available channel that allows higher power secondary transmission). In CRAHNs scenarios, if a SU chooses to use low power for transmission, it may have to rely on multi-hop relaying, whereby each hop has to wait for its own opportunities to transmit [76].

# 3) Operating Frequency

Operating frequency is another key reconfigurable parameter in CRNs. It is the capability of a CR to dynamically reconfigure the CR's centre frequency in response to changes in the RF environment. In [77], a predictive model is proposed to dynamically select the correct configurations, including the operating frequency. This prediction model must be updated continuously to ensure real-time prediction of the correct system configuration to achieve a specified goal based on a set of possible system configurations, environmental conditions, and an expressed demand.

# 4) Channel Bandwidth

Channel bandwidth refers to the width of the spectrum over which a CR transceiver spread its signals. A CR can communicate on either narrower or wider bandwidths, depending on the environment and application. It is crucial for a CR to support variable channel width adaptation in order to operate in heterogeneous networks (i.e. using different wireless technologies). A good example of channel width adaptation is where a Wi-Fi node adapts its channel width to communicate in 5, 10, 20 and 40 MHz channels in order to operate on both 2.4 GHz and TVWS frequencies. Channel bandwidth adaptation can also be useful in rural areas where low throughput links can be tolerated in exchange for increased range and reduced power. This channel width adaptation forms part of reconfiguration capability in CRNs.

#### 5) Communication Technology

The reconfigurable capability of the CR should also allow interoperability among different communication systems (such as cellular GSM, LTE, Wi-Fi, WiMAX, etc.). For instance, a CR can operate under different radio access technologies (RATs) without the need to manually set it. This makes CRNs heterogeneous wireless networks since they are composed of various types of communication technologies and networks.

#### B. Parameter Reconfiguration in CRNs

The centralized CRN, such as a cellular network, can dynamically configure its transceiver with the appropriate RATs and the RF spectrum to adapt to environmental requirements and conditions. Reconfiguration may affect all or most layers of the protocol stack, with the PHY, MAC and Logical Link Control (LLC) and network layers being the most affected. At the PHY and MAC layers, there may be hardware components, such as transceivers, that dynamically change the RATs they operate and the spectrum they use, in order to improve capacity and QoS levels [78]. In this subsection, we look at related work on CR reconfiguration for both centralized and distributed CRNs.

In [79], a heuristic-based two-phase resource allocation scheme is proposed to address the problem of channel and power allocation for SUs in a cellular CRN. This scheme starts by allocating channels and power to the CR BSs to maximize their total coverage area while maintaining the interference constraints for PUs. From there, each BS allocates the channels among the CRs within its cell in order to maximize the total number of CRs served under a particular cell.

Chandra *et al.* [63] made a case for adapting channel widths in Wi-Fi based networks. Properties of different channel widths were studied using measurements from controlled environments. The impact of channel width is characterized based on flow throughput, communication range and power consumption. The results of their experiments approach the theoretical Shannon capacity, for example when the channel width was doubled from 5 to 10 MHz in a certain scenario, the throughout increased by a factor approaching two (1.89 to be precise). They also found that narrower channels are more energy efficient and resilient to multi-path delay spread.

In order to gain a clearer understanding of the impact of CR reconfiguration, the design of experiments (DOE) method is used in [77] to determine how the radio settings affect the CRN performance. DOE is a set of tools and methods for determining cause and effect relationships within a system or process. The authors used the DOE method to determine how parameters at the PHY, data link, network and application layers interact during the reconfiguration.

Kaur *et al.* [80] used fuzzy logic to implement a queuing based adaptive bandwidth allocation scheme in centralized CRNs. (Fuzzy logic is a superset of Boolean logic that has been extended to handle the concept of partial truth.) This scheme assists the channel distributor to decide the quantity of bandwidth that can be allocated to the SUs at any given time.

A hierarchy-based strategy is used to optimize the amount of bandwidth allocated to the SUs depending on the arrival rate of both the PUs and SUs. It is the capability of the scheme to decide on the quantity of bandwidth to be allocated to the SUs, which enables it to reconfigure the SU parameters.

Reference [22] presents a mobile network based functional architecture for the management and control of reconfigurable radio systems. This architecture includes dynamic selforganizing, planning, management, dynamic spectrum management and joint resource management across heterogeneous access technologies.

Kim *et al.* [81] proposed a joint admission control and power allocation scheme for CRAHNs. The proposed scheme allows the SUs to estimate the interference limits at the PU's receiving points depending on the traffic load of the primary network. The authors developed models to analyse the outage probability for SU QoS constraints and the violation probability for PU interference constraints. They use these analytical models to develop a joint admission control and rate/power allocation method subject to QoS and minimum rate requirements, as well as maximum transmit power and fairness constraints for SUs.

A survey of CR as an application of artificial intelligence (AI) with a specific focus on reconfiguration is presented in [82]. In this paper, a cognitive engine (CE) is defined as an intelligent agent responsible for the management of the cognition tasks in a CR. In this context, the term "intelligent" refers to behaviour that is consistent with a specified goal. Using AI, a CE is implemented as an independent entity interacting with the radio transceiver in order to make the decision on how to reconfigure the radio parameters. While current research on CR focuses mainly on DSA, we expect to see the application of AI techniques in CRs to extend beyond DSA to other application areas in the future.

## C. Energy Efficiency

In CRNs, energy is consumed during different spectrum management activities such as spectrum sensing and data reporting [14]. When designing spectrum decision schemes, it is important to consider energy efficient techniques to ensure that less energy is consumed during these cognitive activities. A challenge in CRNs is to strike a balance between the conflicting goals of minimizing the interference to the PU and not compromising the QoS of the SUs [76], [83], [84]. To address this problem, it has been suggested in [83] that the transmit power be adapted based on the reliability of the sensed information. A CR enables us not only to adjust coding, modulation and transmission power, but also to learn and adjust electronic component (such as power amplifier) characteristics in order to minimize energy consumption [85]. Energy consumption is application dependent; therefore there is a need for different energy efficiency models to address the heterogeneous QoS and applications supported by a CRN.

In [85], an energy optimization framework for delay insensitive QoS requirements in CRNs is proposed. This framework enables learning of the radio component characteristics (such as power amplifier efficiency) to adjust radio parameters in order to minimize energy consumption. Based on the CR reconfiguration framework proposed in [85], an intelligent choice of configuration can lead to lower power and energy consumption. In medium- and long-range wireless communications, the power amplifier (PA) usually dominates the system power and energy consumption [85]. Using the CR's reconfiguration capability, it has been shown that energy savings of up to 75% can be achieved compared to conventional systems [85].

A joint optimization solution between the MAC and the PHY layers to maximize energy efficiency is proposed in [84]. This paper investigates energy-efficient transmission duration design and power allocation where a SU selects the available spectrum for data transmission. The link adaptation scheme proposed in [86] balances circuit power consumption and transmission power to achieve the maximum energy efficiency, which is defined as the number of bits transmitted per Joule of energy. The paper demonstrates that energy efficiency can be improved by increasing channel power gain, bandwidth, and by reducing circuit power consumption, which are reconfigurable parameters.

# D. Open Research Issues in CR Reconfiguration

## **1.** Practical Implementation

Most research on parameter reconfiguration in CRNs has been based on computer simulation and mathematical analysis. However, these investigation techniques do not adequately capture all of the technical issues associated with parameter reconfiguration. In order to achieve more realistic and accurate results, it is crucial to build and experiment on more practical implementations of parameter reconfiguration techniques in CRNs.

## 2. Energy Efficiency

In energy constrained scenarios, such as rural areas in Africa or mobile networks, it is important to consider energy efficient approaches during the reconfiguration of the CR parameters. While some aspects of energy efficiency have been explored in existing work, it is still an open challenge to investigate energy efficiency in the context of spectrum decision.

In summary, this section discussed the reconfigurability of CRs in order to satisfy the user's needs. We began by presenting common reconfigurable parameters and then discussed related work in CR reconfiguration. Table III provides a summary of the solutions discussed in this section. We also discussed energy efficiency issues affecting CRNs reconfiguration. We concluded this section by listing open research issues.

# VII. SPECTRUM DECISION IN COGNITIVE RADIO PLATFORMS

In addition to the research efforts presented in the previous sections, there is ongoing work on the development of CR platforms to prototype and implement various techniques on CRNs. A ten year (1999-2009) study of hardware platforms and testbeds related to CR concepts is presented in [87]. There are various software radios and RF hardware available for the

Radio Parame- ters	Existing Techniques	Comments
Modulation and Coding	Adaptive transmission scheme [73].	For centralized CRNs and based on interfer- ence temperature.
Transmission Power	Joint admission control and power allocation scheme [81].	For CRAHNs and SUs estimates the interfer- ence limits at the PUs receiver.
Operating Frequency	Predictive model to dy- namically select the cor- rect operating frequency in [77].	Requires continuous up- date to ensure real-time prediction
Channel Bandwidth	Queuing based adaptive bandwidth allocation scheme [80].	Assists the channel dis- tributor to decide the quantity of bandwidth for SUs in centralized networks.
Communication Technology	Mobile network based reconfigurable radio sys- tems in [22].	Allows CR to operate under different radio ac- cess technologies and includes dynamic self- organizing.

implementation of CR platforms. For a detailed comparison of different CR platforms see [88], where five different platforms are discussed and compared: the Universal Software Radio Peripheral (USRP) hardware [89] and GNU Radio software [90] platform; the Wireless Open Access Research Platform (WARP); the OpenAirInterface.org platform; the Winlab Network Centric Cognitive Radio Hardware Platform (WiNC2R); and the COgnitive Based Radio (COBRA) platform. The next subsections focus on the practical implementation of spectrum decision in CR platforms. Specifically, spectrum characterization. This is because there is little or no work reported in the literature on the practical implementation of spectrum characterization in CR platforms. We conclude this section with open research issues.

## A. Spectrum Selection Implementations

In [30], a channel capacity estimation engine that determines the available capacity of the routes in a CRN is proposed. This engine uses a scheduling aware routing algorithm to choose the best route from all the possible routes by simultaneously considering interference, MAC interactions and estimated capacity.

The GNU Radio based platform is used in [91] for the demonstration of sequence detection algorithms for dynamic spectrum access networks. Channel-sequencing algorithms are normally used to determine the order of multiple non-overlapping channels in which the SU should look for opportunities. A sequence detection algorithm is used for performing spectrum sensing whereby PU's access pattern is exploited using Markov memory modelling techniques. The transition information in the PU channel access is then used for improving spectrum selection in centralized CRNs. In [92], an intelligent SDR platform with CR capabilities [93] is used for the development of context-aware transmission adaptation and channel selection in OFDM-based CRNs using outdoor links

(as opposed to indoor or in-lab links). This work also includes the implementation of spectrum decision functions for longrange rural connectivity provisioning.

#### B. CR Reconfiguration Implementations

Mandke *et al.* [94] developed a CR test-bed called Hydra using GNU Radio software and USRP hardware. The authors used Hydra to perform opportunistic link level rate-adaptation, which demonstrates the reconfiguration capability of a CR. In [95], an implementation of OFDM based systems in a GNU Radio framework is proposed. This platform features a runtime dynamic reconfiguration of radio parameters such as total transmit power, allocated rate and power over sub-channels.

In [96], a dynamic reconfigurable platform based on early access partial reconfiguration (EAPR) is proposed. EAPR is a partial reconfiguration method proposed by Xilinx to enhancement timing performance and simplify the process of building a partial reconfiguration design. This method allows rapid switching of the configuration of radio parameters from one implementation to another without changing the hardware.

In [97], a hardware platform known as Kognitiv networking over white spaces (KNOWS) is proposed for utilizing the TVWS in a Wi-Fi like network. The platform implements the reconfigurable radio using a commercial off-the-shelf (COTS) IEEE 802.11g card to generate the OFDM signals at 2.4 GHz. A wide band frequency synthesizer is used to convert the received signals to the TVWS band. To control the reconfigurable radio, the interface to the MAC layer is a list of register values that specifies the operating frequency, bandwidth, and the transmission power level. The nodes can reconfigure their operating frequency from 400 to 928 MHz in 0.5 MHz steps, as well as the channel bandwidth to 5, 10, 20, and 40 MHz. The threshold for packet reception in the TV band is -85 dBm. The time overhead for adjusting the radio parameters (e.g. frequency, bandwidth, and power level) is less than 100  $\mu$ s.

## C. Open Research Issues in CR Platforms

Over the past decade we have witnessed growing research on CR with interesting theoretical and simulation results published. However, we have seen slower progress on practical hardware and system development for CR technology [87]. This may be associated with different research challenges, three of which are of them are discussed in this subsection.

## **1.** Spectrum Characterization Implementation

The available literature on spectrum decision implementation focuses on spectrum selection and CR reconfiguration. However, to the best of our knowledge, no practical implementations of spectrum characterization on CR platforms have been reported. Therefore, the practical implementation of spectrum characterization in CR platforms remains an open research question that still needs to be addressed.

#### 2. A Complete Functional CR Platform

The majority of CR platforms reported in the literature focus mainly on demonstrating a single aspect of CR (such as spectrum sensing, spectrum selection, or CR reconfiguration) separately. As the field matures, practical test-beds and field trials of the complete dynamic spectrum management framework (1) are needed. In order to achieve this, there is a need to develop and build an integrated CR platform capable of performing all CR functions with an interface to a dynamic geo-location database. More specifically, we would expect to see the practical implementation of integrated spectrum decision incorporating spectrum characterization, spectrum selection and CR reconfiguration.

#### 3. Large-Scale CRN Deployment

Available CR platforms are typically based on a few SDR nodes for research and development purposes. However, there are no existing large-scale outdoor platforms that can be used to study different factors (such as multipath, path loss, the hidden node problem, etc.) affecting CRNs operating as secondary networks within a licensed radio environment. This would be particularly useful in rural and sparsely populated areas. There is therefore a need to develop large-scale CRN platforms to enable real-life CR deployments in the future.

In summary, there are numerous SDR platforms developed to implement and prototype CR functions. GNU Radio software and USRP hardware-based platforms are common because they are cost effective and widely deployed with a large support community. However, the majority of these platforms focus on the implementation of spectrum sensing rather than spectrum decision. In this section, we discussed related work on the implementation of spectrum decision functions on practical CR platforms; and also listed open research issues that still require attention. The next section looks at future developments in CRNs.

#### VIII. FUTURE DEVELOPMENTS IN CRNS

Although the CR area is relatively new, it has grown rapidly in just under a decade. Due to the fact that CRs solve a global pressing need (i.e. how to more efficiently manage scarce and precious RF spectrum), it has received considerable attention from regulatory bodies, standardization bodies, governments, academia and industry around the world. Over the past decade, research on CRs has resulted in thousands of publications in international peer-reviewed journals, magazines, and conference proceedings. Today, almost every call for papers to technical conferences includes at least one topic in CR and spectrum management or efficiency. This growth is expected to continue for at least the next five to ten years and beyond.

This paper covered spectrum decision, one of the key components of CR technology. The open research challenges identified in this paper are focused only on achieving spectrum decision in CRNs. However, looking at CR more broadly as an enabling technology for efficient spectrum management, the future looks bright. This is particularly so with all the ongoing world-wide research and regulatory efforts collectively aimed at making a CR a reality. We now list some of the potential future developments in CRs as an enabling technology for DSA:

- We foresee future developments in CR technology being potentially one of the most influential scientific and engineering endeavours of the 21st century. As indicated in [98], software or computer-based thinking will be the driving force.
- The success of CRs may see it being considered as a key technology in fifth generation (5G) wireless networks.
- Spectrum regulatory agencies across the world will have to face a fundamental paradigm shift in the way they manage the spectrum. In future, network operators may also have a role in regulating their own RF spectrum with the option of leasing to third parties (through the so-called secondary spectrum markets). The beginning of such spectrum reforms are already being witnessed in some countries through spectrum auctions.
- New forms of businesses to facilitate spectrum trade and intermediation promise to dominate in the future. A potential example is the dynamic frequency broker (DFB) system proposed in [99]. DFB acts as a local computerized frequency coordination authority and keeps a complete database of frequency assignments within an area as well as an updated terrain propagation path loss model of its area. It then assigns frequencies on a temporary basis based upon transmitted signal power, spectral density, receive and transmit antenna properties and the required received signal to interference level.
- The integration of AI techniques into CRNs as discussed in [82].

For spectrum decision specifically, we see the following as the potential future developments:

- We are likely going to witness widespread usage of game theoretic approaches as a mathematical tool for addressing spectrum decision and selection problems. This will enable CRs to make autonomous, independent and rational strategic decisions [100] on the available spectrum bands.
- The need for accurate PU characterisation calls for knowledge based on-line databases. Unlike proposed spectrum brokerage business systems, such as in [99], PU activity databases may be controlled by primary network operators, and may only be accessible by CRN operators who are willing to operate as secondary networks.
- Although CR promises efficient utilization of the RF spectrum, it is challenged with network switching delay. This spectrum switching delay includes times for the spectrum decision process in the base station, signalling for establishing new channels, and RF front-end reconfiguration [26]. It is during this switching time that the transmission of the SU is temporarily disconnected which may be detrimental to delay-sensitive applications.
- Interference management and resource allocation is another important topic to consider when dealing with spectrum decision algorithms in CRNs. So, future spectrum decision algorithms should answer this question: How can the SUs transmit simultaneously with the PUs while keeping the interference level of the SUs within an acceptable range [101]?

• Various security issues related to spectrum decision will have to be addressed. For example, how to ensure that only authorised users are permitted to use secondary CRNs and allowed access to sensitive spectrum decision information such as PU and SU activity models.

Finally, more research and development efforts should aim at addressing other relatively unexplored areas of CR-based spectrum management such as spectrum sharing and spectrum mobility. However, the existing research challenges on spectrum sensing and spectrum decision have not been exhausted and doubtless new ones will continue to emerge in the future. These challenges are sure to occupy researchers for some years to come.

# IX. CONCLUSION

Spectrum decision in CRNs is crucial to ensure that appropriate available spectrum bands are selected to satisfy the heterogeneous QoS requirements of the SUs. In this paper, we presented the main functions related to spectrum decision in CRNs based on an extensive study of the existing literature. We explored spectrum decision based on its three key functions: spectrum characterization, spectrum selection and CR reconfiguration. We identified a number of open research issues related to key functions of spectrum decision. We also reviewed various ongoing research work on practical implementations of spectrum decision in CR platforms.

It is worth mentioning that in CRNs, the spectrum decision capability relies on effective spectrum sensing and reliable or accurate data on primary network characteristics (e.g. from geo-location spectrum databases). To realize all of CR's promised capabilities, especially on spectrum decision, the open research issues raised in this paper still need to be addressed. We hope that this paper will be a useful reference point for new and existing researchers in the area of spectrum decision in CRNs, and thus contribute towards the realization of fully functional spectrum aware CRNs.

#### ACKNOWLEDGEMENT

We would like to sincerely thank all of the anonymous reviewers for their constructive comments which have greatly assisted us to improve the quality of our paper.

#### REFERENCES

- Federal Communications Commission, Notice of proposal rulemaking and order, ET Docket No. 03-322, USA, Nov. 2003.
- [2] I. F. Akyildiz, W.-Y. Lee, M. C. Vuran, and S. Mohanty, "NeXt generation/dynamic spectrum access/cognitive radio wireless networks: a survey," *Elsevier Computer Networks*, vol. 50, no. 13, pp. 2127–2159, 2006.
- [3] M. A. McHency, P. A. Tenhula, D. McCloskey, D. A. Reberson, and C. S. Hood, "Chicago spectrum occupancy measurements and analysis and a long- term studies proposal protocol," in *Proc. of 1st Int. Workshop on Technology and Policy for Accessing Spectrum*, Boston, MA, USA, Aug. 5 2006.
- [4] Q. Zhao and B. M. Sadler, "A survey of dynamic spectrum access," *IEEE Signal Processing Magazine*, vol. 24, no. 3, pp. 79–89, May 2007.
- [5] Federal Communications Commission, FCC adopts rules for unlicensed use of television white spaces, Media Release, USA, Accessed: 18/04/2011. [Online]. Available: http://transition.fcc.gov/sptf/headlines2008.html, Nov. 4 2008.

- [6] Office of Communications, Statement on cognitive access to interleaved spectrum, United Kingdom, Accessed: 18/04/2011. [Online]. Available: http://stakeholders.ofcom.org.uk/binaries/consultations/ cognitive/statement/statement.pdf, Jul. 2009.
- [7] M. Nekovee, "A survey of cognitive radio access on TV white spaces," *Hindawi International Journal of Digital Multimedia Broadcasting*, *Article ID: 236568*, pp. 1–11, 2010.
- [8] J. Mitola and G. Q. Maguire, "Cognitive radio: making software radio more personal," *IEEE Personal Communications Magazine*, vol. 6, no. 4, pp. 13–18, Aug. 1999.
- [9] C. E. Shannon, "A mathematical theory of communication," *The Bell System Tech. Journal*, vol. 27, pp. 379–423/623–656, Jul./Oct. 1948.
- [10] C. Santivanez, R. Ramanathan, C. Partridge, R. Krishnan, M. Condell, and S. Polit, "Opportunistic spectrum access: challenges, architecture, protocols," in *Proc. Int. Wireless Internet Conference (WICOM)*, Boston, USA, Aug. 2–5 2006.
- [11] S. Haykin, "Cognitive radio: brain-empowered wireless communications," *IEEE Journal on Selected Areas in Communications*, vol. 23, no. 2, pp. 201–220, 2005.
- [12] M. Fitch, M. Nekovee, S. Kawade, K. Briggs, and R. MacKenzie, "Wireless services provision in TV white space with cognitive radio technology: a telecom operator's perspective and experience," *IEEE Communications Magazine*, vol. 49, no. 3, pp. 64–73, Mar. 2010.
- [13] R. Murty, R. Chandra, T. Moscibroda, and P. Bahl, "SenseLess: A database-driven white spaces network," *IEEE Transactions on Mobile Computing*, vol. 11, no. 2, pp. 189–203, Feb. 2012.
- [14] I. F. Akyildiz, B. F. Lo, and R. Balakrishnan, "Cooperative spectrum sensing in cognitive radio networks: a survey," *Elsevier Personal Communications*, vol. 4, pp. 40–62, 2011.
- [15] C. F. Shih, W. Liao, and H. Chao, "Joint routing and spectrum allocation for multi-hop cognitive radio networks with route robustness consideration," *IEEE Transactions on Wireless Communications*, vol. 10, no. 9, pp. 2940–2949, Sep. 2011.
- [16] S. Ju and J. B. Evans, "Scalable cognitive routing protocol for mobile ad-hoc networks," in *Proc. IEEE GLOBECOM*, Miami, Florida, USA, Dec. 6–10 2010.
- [17] C. Cordeiro, K. Challapali, D. Birru, and S. Sai, "IEEE 802.22: an introduction to the first wireless standard based on cognitive radios," *Academy Journal of Communications*, vol. 1, no. 1, pp. 38–47, Apr. 2006.
- [18] IEEE Std 802.22, Part 22: Cognitive wireless RAN medium access control (MAC) and Physical layer (PHY) specifications: Policies and procedures for operation in the TV bands, IEEE Computer Society, New York, USA, Jul. 2011.
- [19] G. Ko, A. A. Franklin, S. J. You, J. S. Pak, M. S. Song, and C. J. Kim, "Channel management in IEEE 802.22 WRAN systems," *IEEE Communications Magazine*, vol. 48, no. 9, pp. 88–94, Sep. 2010.
- [20] IEEE DySPAN Standards Committee DySPAN-SC, IEEE Std. IEEE DySPAN. [Online]. Available: http://grouper.ieee.org/groups/scc41/
- [21] R. V. Prasad, P. Pawelczak, J. A. Hoffmeyer, and H. S. Berger, "Cognitive functionality in next generation wireless networks: standardization efforts," *IEEE Communications Magazine*, vol. 46, no. 4, pp. 72–78, Apr. 2008.
- [22] M. Mueck, A. Piipponen, K. Kalliojarvi, G. Dimitrakopoulos, K. Tsagkaris, P. Demestichas, F. Casadevall, J. Perez-Romero, O. Sallent, G. Baldini, S. Filin, H. Harad, M. Debbah, T. Haustein, J. Gebert, B. Deschamps, P. Bender, M. Street, S. Kandeepan, J. Lota, and A. Hayar, "ETSI reconfigurable radio systems: Status and future directions on software defined radio and cognitive radio standards," *IEEE Communications Magazine*, vol. 48, no. 9, pp. 78–86, Sep. 2010.
- [23] S. Filin, H. Harada, H. Murakami, and K. Ishizu, "International standardization of cognitive radio systems," *IEEE Communications Magazine*, vol. 49, no. 3, pp. 82–89, Mar. 2011.
- [24] ITU-R Study Group, Working Document towards Draft CPM Text on WRC-12 Agenda Item 1.19, Annex 5 to Document 1B/158, ITU, Accessed: 20/07/2011. [Online]. Available: http://www.itu.int, Feb. 2010.
- [25] ITU-R Study Group, Preliminary draft new report ITU-R [LMS.CRS1]: first report on cognitive radio systems in the land mobile service, ITU, Annex 21 to Document 5S/703-E. [Online]. Available: http://www.itu.int/md/R07-WP5A/new, Jun. 2011.
- [26] W.-Y. Lee and I. F. Akyildiz, "A spectrum decision framework for cognitive radio networks," *IEEE Transactions on Mobile Computing*, vol. 10, no. 2, pp. 161–174, Feb. 2011.
- [27] J. Mitola, "The software radio architecture," *IEEE Communications Magazine*, vol. 33, no. 5, pp. 26–38, May 1995.

- [28] J. Wang, M. Ghosh, and K. Challapali, "Emerging cognitive radio applications: a survey," *IEEE Communications Magazine*, vol. 49, no. 3, pp. 74–81, Mar. 2011.
- [29] M. Hoyhtya, S. Pollin, and A. Mammela, "Improving the performance of cognitive radios through classification, learning, and predictive channel selection," *Advances in Electronics and Telecommunications*, vol. 2, no. 4, pp. 28–38, Dec. 2011.
- [30] V. Kolar, A. Gerco, T. Milcher, M. Petrova, and P. Mahonen, "Capacity estimation and adaptation in cognitive radio networks: Demonstrating software defined radios in action," in *Proc. ACM Int. Conference on Mobile Computing and Networking (MobiCom)*, Beijing, China, Sep. 20–25 2009.
- [31] W.-Y. Lee and I. F. Akyildiz, "Optimal spectrum sensing framework for cognitive radio networks," *IEEE Transactions on Wireless Communications*, vol. 7, no. 10, pp. 3845–3857, 2008.
- [32] G. Wu, S. Singh, and T. Chiueh, "Implementation of dynamic channel switching on ieee 802.11-based wireless mesh networks," in *Proc. International Wireless Internet Conference (WICON)*, Hawaii, USA, Nov. 17–19 2008.
- [33] A. Azarfar, J. F. Frigon, and B. Sanso, "Improving the reliability of wireless networks using cognitive radios," *IEEE Communications Surveys and Tutorials*, vol. 14, no. 2, pp. 338–354, 2012.
- [34] D. Xu, E. Jung, and X. Liu, "Optimal bandwidth selection in multichannel cognitive radio networks: How much is too much?" in *Proc. IEEE Int. Symposium for New Frontiers in DySPAN*, Chicago, Illinois, USA, Oct. 14-17 2008.
- [35] G. Cheng, W. Liu, Y. Li, and W. Cheng, "Joint on-demand routing and spectrum assignment in cognitive radio networks," in *Proc. IEEE Int. Conference on Communications*, Glasgow, Scotland, Jun. 24–28 2007.
- [36] Z. Liang, S. Feng, D. Zhao, and X. Shen, "Delay performance analysis for supporting real-time traffic in a cognitive radio sensor network," *IEEE Transactions on Wireless Communications*, vol. 10, no. 1, pp. 325–335, Jan. 2011.
- [37] M. Amir, A. El-Keyi, and M. Nafie, "Constrained interference alignment and the spatial degrees of freedom of mimo cognitive networks," *IEEE Transactions on Information Theory*, vol. 57, no. 5, pp. 2994– 3004, May 2011.
- [38] S. M. Almalfouh and G. L. Stber, "Interference-aware radio resource allocation in OFDMA-based cognitive radio networks," *IEEE Transactions on Vehiclular Technology*, vol. 60, no. 4, pp. 1999–3004, May 2011.
- [39] H. Bogucka, A. M. Wyglinski, S. Pagadarai, and A. Kliks, "Spectrally agile multicarrier waveforms for opportunistic wireless access," *IEEE Communications Magazine*, vol. 49, no. 6, pp. 108–115, Jun. 2011.
- [40] T. Li, W. Mow, V. Lau, M. Siu, R. S. Cheng, and R. D. Murch, "Robust joint interference detection and decoding for OFDM-based cognitive radio systems with unknown interference," *IEEE Journal on Selected Areas in Communications*, vol. 25, no. 3, pp. 566–575, Apr. 2007.
- [41] A. Rabbachin, T. Q. S. Quek, H. Shin, and M. Z. Win, "Cognitive network interference," *IEEE Journal of Selected Areas in Communications*, vol. 29, no. 2, pp. 480–493, Feb. 2011.
- [42] V. Pla, J. R. Vidal, J. Martinez-Bauset, and L. Guijarro, "Modeling and characterization of spectrum white spaces for underlay cognitive radio networks," in *Proc. IEEE Int. Conference on Communications (ICC)*, Cape Town, South Africa, May 23–27 2010.
- [43] Y. Yuan, P. Bahl, R. Chandra, T. Moscibroda, and Y. Wu, "Allocating dynamic time-spectrum blocks in cognitive radio networks," in *Proc. ACM International Symposium on Mobile ad hoc Networking and Computing Wireless Communications and Networking Conference*, Quebec, Canada, Sep. 9–14 2007.
- [44] M. Matinmikko, M. Hoyhtya, M. Mustonen, H. Sarvanko, A. Hekkala, M. Katz, A. Mammela, M. Kiviranta, and A. Kautio, "Cognitive radio: an intelligent wireless communication system," Tech. Rep., 2008.
- [45] H. A. Suraweera, P. J. Smith, and M. Shafi, "Capacity limits and performance analysis of cognitive radio with imperfect channel knowledge," *IEEE Transactions on Vehicular Technology*, vol. 59, no. 4, pp. 1811– 1822, May 2010.
- [46] A. K. Kaur and A. K. Sharma, "BER performance analysis of cognitive radio physical layer over rayleigh fading channel," *International Journal of Computer Applications*, vol. 25, no. 11, pp. 25–29, Jul. 2011.
- [47] Working Group on Wireless Regional Area Networks, IEEE Std. 802.22, 2011. [Online]. Available: http://www.ieee802.org/22/
- [48] T. S. Rappaport, *Wireless Communications: Principles and Practice*. 2nd ed. Prentice-Hal, 2002.
- [49] M. B. Pursley and T. C. Royster, "Low-complexity adaptive transmission for cognitive radios in dynamic spectrum access networks," *IEEE*

Journal on Selected Areas in Communications, vol. 26, no. 1, pp. 83–94, Jan. 2008.

- [50] D. A. Roberson, C. S. Hood, J. L. LoCicero, and J. T. MacDonald, "Spectrum occupancy and interference studies in support of cognitive radio technology deployment," in *Proc. 1st IEEE Workshop on Networking Technology for Software Defined Radio*, Reston, Virginia, USA, Sep. 26–35 2006.
- [51] Q. Zhao, L. Tong, and A. Swami, "Decentralized cognitive MAC for dynamic spectrum access," in *Proc. 1st IEEE Int. Symposium for New Frontiers in DySPAN*, Reston, Virginia, USA, Sep. 26–35 2005.
- [52] B. Canberk, I. F. Akyildiz, and S. Oktug, "Primary user activity modeling using first-difference filter clustering and correlation in cognitive radio networks," *IEEE/ACM Transactions on Networking*, vol. 19, no. 1, pp. 170–183, Feb. 2011.
- [53] C. Chou, S. Shankar, H. Kim, and K. G. Shin, "What and how much to gain by spectrum agility," *IEEE Journal on Selected Areas* in Communications, vol. 25, no. 3, pp. 576–588, Apr. 2007.
- [54] A. A. Daoud, M. Alanyali, and D. Starobinski, "Secondary pricing of spectrum in cellular CDMA networks," in *Proc. IEEE Int. Symposium* for New Frontiers in DySPAN, Dublin, Ireland, Apr. 17–20 2007.
- [55] D. Willkomm, S. Machiraju, J. Bolot, and A. Wolisz, "Primary users in cellular networks: a large-scale measurement study," in *Proc. IEEE Int. Symposium for New Frontiers in DySPAN*, Chicago, Illinois, USA, Oct. 14-17 2008.
- [56] V. Paxson and S. Floyd, "Wide-area traffic: the failure of poison modelling," *IEEE/ACM Transactions on Networking*, vol. 3, no. 3, pp. 226–244, 1995.
- [57] P. A. K. Acharya, S. Singh, and H. Zheng, "Reliable open spectrum communications through proactive spectrum access," in *Proc. ACM* 66th Int. Workshop on Technology and Policy for Accessing Spectrum, Boston, USA, Aug. 2–5 2006.
- [58] S. Yarkan and H. Arslan, "Binary time series approach to spectrum prediction for cognitive radio," in *Proc. IEEE Vehicular Technology Conference*, Baltimore, USA, Sep./Oct. 30–3 2007.
- [59] J. Riihijarvi, P. Mahonene, M. Wellens, and M. Gordziel, "Characterization and modelling of spectrum for dynamic spectrum access with spatial statistics and random fields," in *Proc. IEEE 19th Int. Symposium* on *Personal, Indoor and Mobile Radio Communications*, Cannes, Sep. 15–18 2008.
- [60] P. A. P. Moran, "Notes on continuous stochastic phenomena," *Biometrika*, vol. 37, no. 1, pp. 17–23, 1950.
- [61] R. C. Geary, "The contiguity ratio and statistical mapping," *The Incorporated Statistician*, vol. 5, no. 3, pp. 115–145, Nov. 1954.
- [62] C. Ghosh, S. Pagadarai, D. Agrawal, and A. Wyglinski, "A framework for statistical wireless spectrum occupancy modeling," *IEEE Transactions on Wireless Communications*, vol. 9, no. 1, pp. 38–44, Jan. 2010.
- [63] R. Chandra, R. Mahajan, T. Moscibroda, R. Raghavendra, and P. Bahl, "A case for adapting channel width in wireless networks," in *Proc. ACM SIGCOMM Conference on Applications, Technologies, Architectures, and Protocols for Computer Communications*, Seattle, USA, Aug. 17–22 2008.
- [64] P. Bahl, R. Chandra, T. Moscibroda, and R. Murty, "White space networking with Wi-Fi like connectivity," in *Proc. ACM SIGCOMM Conference on Applications, Technologies, Architectures, and Protocols* for Computer Communications, Barcelona, Spain, Aug. 17–21 2009.
- [65] I. Malanchini, M. Cesana, and N. Gatti, "On spectrum selection games in cognitive radio networks," in *Proc. IEEE GLOBECOM*, Honolulu, Hawaii, USA, Nov./Dec. 30–4 2009.
- [66] D. Na, H. Nan, and S. Yoo, "Policy-based dynamic channel selection architecture for cognitive radio networks," in *Proc. IEEE Int. Conference on Communications and Networking in China*, Shanghai, China, Aug. 22–24 2007.
- [67] I. F. Akyildiz, W.-Y. Lee, and K. R. Chowdhury, "CRAHNS: cognitive radio ad hoc networks," *Elsevier Journal on Ad Hoc Networks*, vol. 7, pp. 810–826, 2009.
- [68] H. Khalife, S. Ahuja, N. Malouch, and M. Krunz, "Joint routing and spectrum selection for multihop cognitive radio networks," Tech. Rep., 2007. [Online]. Available: http://www-rp.lip6.fr/khalife/tech-report.pdf, 2007.
- [69] Q. Wang and H. Zheng, "Route and spectrum selection in dynamic spectrum networks," in *Proc. IEEE 3rd Consumer Communications* and Networking Conference, Las Vegas, USA, Jan. 8–10 2006.
- [70] H. Khalife, S. Ahuja, N. Malouch, and M. Krunz, "Probabilistic path selection in opportunistic cognitive radio networks," in *Proc. IEEE GLOBECOM*, New Orleans, USA, Nov./Dec. 30–4 2008.
- [71] L. Ding, T. Melodia, S. N. Batalama, J. D. Matyjas, and M. J. Medley, "Cross-layer routing and dynamic spectrum allocation in cognitive"

radio ad hoc networks," *IEEE Transactions on Vehicular Technology*, vol. 54, no. 4, pp. 1969–1979, May 2010.

- [72] S. Sengupta and K. P. Subbalakshmi, "Open research issues in multihop cognitive radio networks," *To appear in IEEE Communications Magazine*, 2012.
- [73] M. Hong, J. Kim, and Y. Shin, "An adaptive transmission scheme for cognitive radio systems based on interference temperature model," in *Proc. IEEE Consumer Communications and Networking Conference*, Las Vegas, USA, Jan. 10–12 2008.
- [74] Z. Ha and K. J. R. Liu, Resource allocation for wireless networks: basics, techniques, and applications, 1st ed. New York, USA: Cambridge University Press, 2008.
- [75] Y. Che, J. Wang, J. Chen, W. Tang, and S. Li, "Hybrid power control scheme in hierarchical spectrum sharing network for cognitive radio," *Elsevier Physical Communications*, vol. 2, no. 1-2, pp. 73–86, Mar-Jun. 2009.
- [76] W. Ren, Q. Zhao, and A. Swami, "Power control in spectrum overlay networks: how to cross a multi-lane highway," in *Proc. IEEE Int. Conference on Acoustic, Speech and Signal Processing*, Las Vegas, USA, Mar./Apr. 31–4 2008.
- [77] T. Weingart, D. C. Sicker, and D. Grunwald, "A statistical method for reconfiguration of cognitive radios," *IEEE Wireless Communications*, vol. 14, no. 4, pp. 34–40, Aug. 2007.
- [78] P. Demestichas, G. Dimitrakopoulos, K. Tsagkaris, K. Demestichas, and J. Adamopoulou, "Reconfigurations selection in cognitive, beyond 3G, radio infrastructures," in *Proc. Int. Conference on Cognitive Radio Oriented Wireless Networks and Communications (CROWNCOM)*, Mykonos Island, Greece, Jun. 8–10 2006.
- [79] A. T. Hoang and Y.-C. Liang, "A two-phase channel and power allocation scheme for cognitive radio networks," in *Proc. IEEE Int. Symposium on Personal, Indoor and Mobile Radio Communications*, Helsinki, Finland, Sep. 11-14 2006.
- [80] P. Kaur, M. Uddin, and A. Khosla, "Adaptive bandwidth allocation scheme for cognitive radios," *International Journal of Advancements* in Computing Technology, vol. 2, no. 2, pp. 35–41, Jun. 2010.
- [81] D. I. Kim, L. B. Le, and E. Hossain, "Joint rate and power allocation for cognitive radios in dynamic spectrum access environment," *IEEE Trans. on Wireless Communications*, vol. 7, no. 12, pp. 5517–5527, Dec. 2008.
- [82] A. He, K. K. Bae, T. R. Newman, J. Gaeddert, K. Kim, R. Menon, L. Morales-Tirado, J. Neel, Y. Zhao, R. J. H, and W. H. Tranter, "A survey of artificial intelligence for cognitive radios," *IEEE Transactions* on Vehicular Technology, vol. 59, no. 4, pp. 1578–1592, May 2010.
- [83] S. Srinivasa and S. A. Jafar, "Soft sensing and optimal power control for cognitive radio," in *Proc. IEEE Global Telecoms Conference*, Washington D.C., USA, Nov. 26–30 2007.
- [84] L. Li, X. Zhou, H. Xu, G. Y. Li, D. Wang, and A. Soong, "Energyefficient transmission in cognitive radio networks," in *Proc. IEEE Consumer Comm. and Networking Conference*, Las Vegas, USA, Jan. 9–12 2010.
- [85] A. He, S. Srikanteswara, J. H. Reed, X. Chen, W. H. Tranter, K. K. Bae, and M. Sajadieh, "Minimizing energy consumption using cognitive radio," in *Proc. IEEE Int. Performance Computing and Communications Conference*, Austin, Texas, USA, Dec. 7–9 2008.
- [86] G. Miao, N. Himayat, and G. Y. Li, "Energy-efficient link adaptation in frequency selection channels," *IEEE Transactions on Communications*, vol. 57, no. 2, pp. 545–554, Feb. 2010.
- [87] P. Pawelczak, K. Nolan, L. Doyle, S. W. Oh, and D. Cabric, "Cognitive radio: ten years of experimentation and demonstration," *IEEE Communications Magazine*, vol. 49, no. 3, pp. 90–100, Mar. 2011.
- [88] O. Gustafsson, K. Amiri, D. Andersson, A. Blad, C. Bonnet, J. R. Cavallaro, J. Declerck, A. Dejonghe, P. Eliardsson, M. Glasse, A. Hayar, L. Hollevoet, C. Hunter, M. Joshi, F. Kaltenberger, R. Knopp, K. Le, Z. Miljanic, P. Murphy, F. Naessens, N. Nikaein, D. Nussbaum, R. Pacalet, P. Raghavan, A. Sabharwal, O. Sarode, P. Spasojevic, S. Yang, H. M. Tullberg, T. Vander-Aa, L. Van der Perre, M. Wetterwald, and M. Wu, "Architectures for cognitive radio testbeds and demonstrators an overview," in *Proc. Int. Conference on Cognitive Radio Oriented Wireless Networks and Communications (CROWNCOM)*, Cannes, Jun. 9–11 2010.
- [89] Universal software radio peripheral (USRP). Ettus Research. [Online]. Available: http://www.ettus.com/
- [90] GNU radio website. [Online]. Available: http://gnuradio.org/redmine/
- [91] Z. Sun, G. Bradford, and J. N. Laneman, "Demonstration of sequence detection algorithms for dynamic spectrum access networks," in *Proc. IEEE Int. Symposium for New Frontiers in DySPAN*, Singapore, Apr. 6–9 2010.

- [92] V. Pejovic and E. Belding, "A context-aware approach to wireless transmission adaptation," in Proc. IEEE Communications Society Conference on Sensor, Mesh and Ad Hoc Communications and Networks (SECON), Utah, Salt Lake City, USA, Jun. 27-30 2011.
- [93] M. T. Masonta, D. Johnson, and M. Mzyece, The White Space Opportunity in Southern Africa: Measurements with Meraka Cognitive Radio Platform, R. Popescu-Zeletin, K. Jonas, I. A. Rai, R. Glitho, and A. Villafiorita, Eds. Lecture Notes of the Institute for Computer Sciences, Social Informatics and Telecommunications Engineering (LNICST), Part 1, Volume 92, e-Infrastructure and e-Services for Developing Countries, pp. 64-73, Feb. 2012.
- [94] K. Mandke, S. Choi, G. Kim, R. Grant, R. C. Daniels, W. Kim, R. W. Heath, and S. Nettles, "Early results on hydra: a flexible MAC/PHY multihop testbed," in Proc. IEEE Vehicular Technology Conference, Dublin, Ireland, Apr. 23-25 2007.
- [95] M. Zivkovic, D. Auras, and R. Mathar, "OFMD-based dynamic spectrum access," in Proc. IEEE Int. Symposium for New Frontiers in DySPAN, Singapore, Apr. 6-9 2010.
- [96] K. G. Nezami, P. W. Stephen, and S. D. Walker, "Handel-C implementation of early-access partial reconfiguration for software defined radio," in Proc. IEEE Wireless Communications and Networking Conference, Las Vegas, USA, Mar./Apr. 31-3 2008.
- [97] Y. Yuan, P. Bahl, R. Chandra, P. A. Chou, J. I. Ferrell, T. Moscibroda, S. Narlanka, and Y. Wu, "KNOWS: cognitive radio networks over white spaces," in Proc. 2nd IEEE Int. Symposium for New Frontiers in DySPAN, Dublin, Ireland, Apr. 17-20 2007.
- [98] S. Haykin, "Cognitive radio: research challenges," in Proc. IEEE Vehicular Technology Conference, Calgary, Canada, Sep. 21-24 2008.
- [99] T. Maseng and T. Ulversory, "Dynamic frequency broker and cognitive radio," in Proc. IET Seminar on Cognitive Radio and Software Defined Radio: Technology and Techniques, London, UK, Sep. 18 2008.
- [100] W. Saad, Z. Han, M. Debbah, A. Hjorungnes, and T. Basar, "Coalitional game theory for communication networks: a tutorial," IEEE Signal Processing Magazine, vol. 26, no. 5, pp. 77-97, Sep. 2009.
- [101] A. M. Hayar, R. Knopp, and R. Pacalet, "Cognitive radio research and implementation challenges," in Proc. IEEE Asilomar Conference on Signals, Systems and Computers, Pacific Grove, USA, Nov. 4-7 2007.



Moshe Timothy Masonta (S'2010) is a Senior Researcher at the Council for Scientific and Industrial Research Meraka Unit, Pretoria and a Doctorate candidate in Electrical Engineering (Telecommunications technology), at Tshwane University of Technology (TUT), Pretoria, South Africa. He received a Masters in Technology degree (2008) from TUT, and an MSc in Electronic Engineering degree (2010) from Ecole Suprieure dIngnieurs en Electrotechnique et Electronique (ESIEE) de Paris, France.

From September to December 2011 he was a visiting student at the University of California, Santa Barbara. He is a member of the European Cooperation on Science and Technology (COST) Action IC0905 TERRAs Special Interest Group on energy efficiency. His research interests are in dynamic spectrum access and management, cognitive radio systems, television white space spectrum, spectrum regulations and energy efficiency in wireless networks.



Mjumo Mzyece (S'1999 - M'2004) is an associate professor at the French South African Institute of Technology (F'SATI) in the Department of Electrical Engineering at Tshwane University of Technology (TUT) in Pretoria, South Africa, where he is engaged in research and teaching on communications-related themes. His research interests are in next-generation wireless networks, including those based on dynamic spectrum access technologies; the design and evaluation of communications protocols; and the interaction of technological and other critical factors (e.g. policy, regulatory, commercial/economic, etc.).

Outside of the academic world, he has held various technical, management and consulting positions locally and internationally. He holds a BEng (Honours) degree in Electronic & Electrical Engineering from the University of Manchester (England), an MSc (with Distinction) in Communications Technology & Policy and a PhD in Electronic & Electrical Engineering, both from the University of Strathclyde (Scotland).



Ntsibane Ntlatlapa (M'2007 - SM'2010) is a manager of the Networks and Media competency area at CSIR Meraka Institute. The competency area carries out research on wireless communications and networking, real-time media streaming and localization. The broad objectives of the activities in his competency area are to develop core components of technologies aimed at bringing Broadband Internet access and digital services to rural communities and to solve communication challenges in harsh environments such as underground mines. His research

interests cover several sub-fields of networking, including wireless mesh and sensor networks protocol optimization.

Dr Ntlatlapa obtained BSc, MSc and PhD in Computer Science from the National University of Lesotho, University of East Anglia (Norwich, UK) and Auburn University (Alabama, USA) respectively. Prior to joining the CSIR, he worked as a Computer Science lecturer at the National University of Lesotho. While there, he held a number of positions in the faculty, including two years as Head of Department of Mathematics and Computer Science. During that period, he also worked as a senior consultant to the Government of Lesotho in the run-up to the first phase of the World Summit on Information Society, which was held in Geneva in 2003.

Dr Ntlatlapa has 20 years work experience in academia. He has lectured at both undergraduate and postgraduate levels at the National University of Lesotho, Alabama State University, Auburn University (both in the USA) and at the University of Part Harcourt (Nigeria). He has also supervised students at Tshwane University of Technology and the University of Zululand (both in South Africa).